

### Sick Happens: The Effect of Worker Health Shocks on Coworkers' Employment and Health Behavior

by

Wolfgang FRIMMEL Rene WIESINGER

Working Paper No. 2412 October 2024

> Johannes Kepler University of Linz Department of Economics Altenberger Strasse 69 A-4040 Linz - Auhof, Austria www.econ.jku.at

> > wolfgang.frimmel@jku.at

# Sick Happens: The Effect of Worker Health Shocks on Coworkers' Employment and Health Behavior<sup>\*</sup>

Wolfgang Frimmel<sup>a</sup> and Rene Wiesinger<sup>b</sup>

<sup>a</sup>Department of Economics, Johannes Kepler University Linz <sup>b</sup>Parliamentary Budget Office

October 11, 2024

#### Abstract

We analyze how a worker's severe health shock affects the employment and health behavior of their older coworkers. We link comprehensive administrative data on labor market histories and health records from Austria to identify coworker networks and severe health shocks in small firms, which cause substantial increases in healthcare expenditures, absenteeism, and mortality, as well as persistent reductions in the labor supply of affected workers. Combining a matching approach with a difference-in-difference framework, we find a significant impact of a health shock on the labor market outcomes and health behavior of older coworkers. Affected coworkers are about 2.3 percentage points more likely to be employed in the shock firm and tend to delay retirement. Although there is no change in daily earnings and earnings growth, coworkers are more likely to receive special bonus payments after leaving the firm. The employment effects are larger when the health shock affects a high-skilled worker and when the shocked worker leaves the firm after the health shock. Finally, we find that female coworkers in the treatment group are more likely to have a mammography, especially in response to health shocks due to cancer. We find no statistically significant effects on participation in general health check-ups and PSA tests, or on coworker absenteeism.

JEL Classification: I10 · I12 · J20 · J21 Keywords: coworker health shock · employment · retirement · health behavior · difference-in-difference

<sup>\*</sup>Frimmel: Department of Economics, Johannes Kepler University Linz, Altenberger Straße 69, A-4040 Linz, Austria (email: wolfgang.frimmel@jku.at). Wiesinger: Parliamentary Budget Office, Stubenring 8-10, A-1010 Vienna, Austria (email: rene.wiesinger@parlament.gv.at). For helpful discussions and comments we thank Felix Glaser. Wiesinger would especially like to thank his father, who provided the inspiration for this paper by sharing one of his anecdotes. Raneem Elhifnawy provided helpful research assistance.

## 1 Introduction

Every year, millions of individuals around the world suffer a major health shock. In 2021 alone, an estimated 66.8 million cases of cardiovascular diseases and 23.6 million cases of cancer were recorded. Both diseases are among the top three causes of death.<sup>1</sup> While the far-reaching and detrimental consequences of such health shocks are well documented for the affected individuals (see Prinz et al. 2018 for a literature review), there is less evidence on the spillover effects of such events on those surrounding them. Existing studies mainly focus on family members (e.g., Fadlon and Nielsen, 2019; Frimmel et al., forthcoming), while networks outside the household, such as coworkers, have received little attention. However, adults spend a significant amount of their time at work. In the US, individuals aged between 20 and 64 spend an average of 2.9 hours per day with their coworkers. This is about the same amount spent with their partner or children, and considerably less than the time spent with the extended family or friends.<sup>2,3</sup>

A health shock of a worker can therefore also affect their coworkers in several dimensions. A priori, the effect of a severe health shock on coworkers' employment is ambiguous. Witnessing a major deterioration in a worker's health makes coworkers' own health risks more salient (Fadlon and Nielsen, 2019). Worried about their own health and that the health shock could be attributed to occupational risk and workload, coworkers may try to reduce their own risk of suffering a severe health shock by reducing their labor supply, switching to less demanding jobs, or retiring earlier. In this case, a health shock is expected to have a negative effect on coworkers' employment and earnings and to lead to a higher probability of (early) retirement. At the same time coworkers may adjust their preventive health behavior accordingly. Coworkers may be more likely to participate in health screening programs to learn about their own health risks, or they may try to reduce risky health behaviors. Alternatively, a worker's health may affect the labor supply of coworkers through firm-level effects. A deterioration in a worker's health may result in a partial loss of human capital and productivity. Firms may try to compensate for reduced productivity either by hiring new external workers or by relying more on existing workers. If these two groups are imperfect substitutes, firms' demand for the labor of incumbent workers should increase (Jäger and Heining, 2022), resulting in increased employment of coworkers (at least in the shock firm) or delayed retirement.

In this study, we analyze how a worker's severe health shock affects the labor market outcomes and health behavior of their coworkers. We link comprehensive administrative data on labor market histories and health records from Austria to identify coworker net-

<sup>&</sup>lt;sup>1</sup> Source: Institute for Health Metrics and Evaluation: Cardiovascular Diseases, Cancer.

<sup>&</sup>lt;sup>2</sup> The values are 2.9 hours per day spent with a partner, 2.5 hours with children, 1.2 hours with extended family, and 0.8 hours with friends. Source: Our World in Data, own calculations.

<sup>&</sup>lt;sup>3</sup> Time use data from Austria suggest that the average day of a 20 to 64 year old includes 4.1 hours of work, 3 hours of care work, and 5 hours of social contacts and leisure time (Statistik Austria, 2023, own calculations).

works and severe health shocks in small firms. Health shocks are defined as a worker's first hospitalization for cancer or acute cardiovascular events, including stroke, myocardial infarction, and heart failure. We show that these health shocks are indeed severe, leading to substantial increases in healthcare expenditures, absenteeism, and mortality, as well as to significant and persistent reductions in labor supply for those who suffer them. To construct an appropriate control group, we follow the method implemented by Jäger and Heining (2022). For each worker-firm pair with an actual shock we use a combination of exact and propensity score matching to find a worker-firm pair in the pool of worker-firm pairs that do not experience a health shock and add information on coworkers. Using a difference-in-difference framework, we compare coworkers who experience a severe health shock of a colleague to coworkers without such an experience, but who have a colleague who is similar to an actual shock worker on key observable characteristics, before and after the (placebo) health shock. For our main analysis, we focus on coworkers aged 50 or older at the time of the health shock. This group may be particularly affected because the likelihood of experiencing a severe health shock increases with age, and thus learning about one's own health risks may be greater for more similar older coworkers (Fadlon and Nielsen, 2019). Moreover, firms may rely on seniority to find reliable replacements for shock workers (Bianchi et al., 2023).

We find a significant impact of a health shock on older coworkers' labor market outcomes and health behavior. Coworkers in the treatment group are 2.3 percentage points more likely to be employed in the shock firm after the health shock. This corresponds to an increase of 2.5 percent relative to the pre-shock employment probability. Consistently, employment days increase and retirement days decrease. Although there is no change in daily earnings and earnings growth, treated coworkers are more likely to receive special severance payments (i.e., golden handshakes) when they leave the firm. The effects on employment and retirement are larger when the health shock affects a high-skilled worker and when the shock worker leaves the firm after the health shock. Regarding health behavior, we find that female coworkers in the treatment group are 9.7 percent more likely to have a mammography after the health shock than women in the control group. This effect appears to be larger when coworkers and shock workers are more similar to each other in terms of age and gender. Since mammographies are used to detect breast cancer, we find that the increase is mainly driven by cancer-related health shocks. We find no statistically significant effects on participation in general health check-ups and prostatespecific antigen (PSA) tests. Coworker absenteeism is also unaffected. While these results all focus on coworkers who are over 50 at the time of the health shock, we find no or only considerably smaller effects for younger coworkers.

We contribute to several strands of the literature. First, we extend the largely grown literature on the spillover effects of health shocks. Existing studies have mainly focused on the family, in particular on the effects on household wealth (e.g., Wu, 2003), spouses' labor supply (e.g., Coile, 2004; Fadlon and Nielsen, 2021; García-Gómez et al., 2013; Jeon and Pohl, 2017; Nahum, 2007), and well-being (e.g., Angelini and Costa-Font, 2023).<sup>4</sup> There is considerably less evidence on spillovers to groups outside the household, such as coworkers. One exception to this is the paper by Fadlon and Nielsen (2019), which is based on administrative data from Denmark. Although the authors are primarily interested in the effect of health shocks on spouses' and children's health behavior, they find that the increase in preventive care associated with a health shock also extends to coworkers. They show that the results are driven by coworkers who are similar to the shock worker in terms of age and occupation class. Similarly, Kapeller (2021) uses data from Austria to show that a worker's cancer diagnosis leads coworkers who are similar to the shock worker to participate more frequently in health screening programs, such as general health check-ups, PSA tests, and gynecological check-ups. These effects even cascade to coworkers' spouses. Pruckner et al. (2020) find evidence that worker health behaviors (i.e., the participation in different forms of health screening) are highly correlated. This is particularly true when workers have similar characteristics.

Second, we extend the literature on the labor market response of coworkers to health shocks, where the study by Jäger and Heining (2022) is most closely related to our work. They use the death of a worker to estimate the magnitude of personnel replacement cost for firms in the German labor market.<sup>5</sup> Their results suggest that incumbent workers and external hires are not perfect substitutes, as the death of a worker increases the earnings of coworkers and the probability of being employed by the same firm. At the same time, the probability of employment in other firms decreases. Larger effects emerge when the deceased worker has more specific human capital and when there are only few workers in the deceased worker's occupation in the local external labor market.<sup>6</sup> Using administrative data from Brazil, Fietz and Schmeißer (2024) show how changes in the racial composition of a firm's workforce caused by worker deaths affect the retention of employees. They find that, relative to the death of a white worker, the loss of a non-white worker reduces the probability of remaining at the firm by 1.8 percent for non-white coworkers. Since the effect is driven by quits, this suggests that the change in the probability of staying is due to changes in labor supply rather than changes in labor demand. Other studies using deaths as an exogenous variation in worker exits focus on spillover effects in very specific settings, such as the effect of CEOs on firm performance (e.g., Becker and Hvide, 2022)

<sup>&</sup>lt;sup>4</sup> Several studies also document the effects of parental health shocks on (adult) children's labor supply and/or well-being (e.g., Frimmel et al., forthcoming; Glaser and Pruckner, 2023; Maestas et al., 2024; Norén, 2020; Ramirez Lizardi et al., 2024; Rellstab et al., 2020).

<sup>&</sup>lt;sup>5</sup> Bertheau et al. (2022) also use worker deaths to estimate replacement costs. They find that worker exits due to unexpected deaths reduce the profits of Danish firms by a total of 41,000 Euros. This profit loss is mainly due to a loss of value added and an increase in incumbent workers' earnings.

<sup>&</sup>lt;sup>6</sup> Spillover effects on coworkers' careers have also been found for changes in labor supply that are not related to workers' deaths. For example, Bianchi et al. (2023) study an Italian pension reform to show that the delayed retirement leads to lower wage growth and promotion rates for coworkers. The effects are larger when senior workers are in a higher position and for coworkers over the age of 55.

and the effect of individual inventors and superstar academics on the success of their team members (e.g., Azoulay et al., 2010; Jaravel et al., 2018; Oettl, 2012).

Finally, we also complement the literature on peer effects on different dimensions of individual behavior. Studies in this area have estimated peer effects in welfare use and participation in social security programs (e.g., Åslund and Frederiksson, 2009; Dahl et al., 2014; Markussen and Røed, 2015), risky health behavior (e.g., Eisenberg et al., 2014; Fletcher, 2012; Gaviria and Raphael, 2001), preventive healthcare (e.g., Bouckaert et al., 2020; Francetic et al., 2022; Goldberg et al., 2023; Redler and Reichel, 2024), worker absenteeism (e.g., Bradley et al., 2007; Hesselius et al., 2009; Ichino and Maggi, 2000), productivity (e.g., Bandiera et al., 2010; Guryan et al., 2009; Mas and Moretti, 2009), and wages (e.g., Cornelissen et al., 2017). Our results on the negative impact of severe health shocks on shock workers' own employment also add to the existing evidence on the adverse employment and earnings consequences of health shocks for those experiencing them (e.g., Ahammer et al., 2024; Bíró et al., 2024; Dobkin et al., 2018; García-Gómez, 2011; Halla and Zweimüller, 2013; Parro and Pohl, 2021; Trevisan and Zantomio, 2016).

Our study contributes to the existing literature in three ways. First, we extend the existing evidence by considering the impact of severe health shocks on coworkers' labor market outcomes and health behavior. Other studies mainly focus on worker deaths, especially those interested in the labor market responses of coworkers. We believe that health shocks are a more relevant treatment than deaths because they occur more frequently and therefore represent a more realistic scenario for a wide range of firms. Second, we consider responses to an event that mainly leads to a partial loss of productivity and human capital. In the case of worker deaths, firms face a total loss of productivity, which is not necessarily the case for severe health shocks. Firms and coworkers may react differently to a partial and/or temporary substitution of shock workers than to a full and/or permanent substitution. Finally, we synthesize the two strands of the literature that consider the spillover effects of health shocks on health behavior and labor market responses. Our detailed and comprehensive administrative data on individual-level labor market activity and healthcare utilization allow us not only to study changes in coworkers' labor market outcomes, but also to consider healthcare utilization and preventive health behavior as relevant outcomes of interest.

The remainder of the paper is organized as follows. Section 2 summarizes the institutional setting of this study and describes the data used in the empirical analysis. Section 3 provides a detailed description of the empirical strategy. Section 4 presents our main results, effect heterogeneity, and different robustness checks. Section 5 concludes the paper.

# 2 Institutional Setting and Data

### 2.1 Institutional Setting

Austrian Labor Market: The Austrian labor market is characterized by a combination of strong institutional regulation of wages and working conditions and relative flexibility in terms of dismissals and turnover. Wages are generally set by collective agreements negotiated between union and employer representatives (Pollan, 2009). About 98 percent of the Austrian workforce is covered by a collective bargaining agreement.<sup>7</sup> At the same time, turnover is relatively high and employment protection against dismissal is relatively weak (Böheim, 2017). Firms can unilaterally dismiss workers without having to give a specific reason, subject to a statutory notice period.

The Austrian economy is characterized by small firms rather than large corporations. Our analysis focuses on firms that employ fewer than 50 workers four quarters before the health shock. In 2021, firms with a firm size between 1 and 49 workers accounted for about 36 percent of all firms. Together, they employed 34 percent of all workers.<sup>8</sup> Our sample therefore covers a non-negligible part of the Austrian labor market.

Social Security: Austria has a Bismarckian social security system that offers universal access to healthcare, pensions, disability, and unemployment benefits. Individuals are assigned to one of the health insurance funds based on their place of residence and occupation. The Austrian Health Insurance Fund (Österreichische Gesundheitskasse) is the largest insurer, covering about three-quarters of all private-sector workers.<sup>9</sup> Health insurance provides access to a wide range of healthcare services, including outpatient visits, inpatient care, prescription drugs, and health and cancer screening programs.<sup>10</sup> Expenditures on outpatient care and medication are financed mainly by wage-based social security contributions paid by employees and employers. Hospital costs are also partly financed by tax revenues. General practitioners (GPs) have traditionally been the first point of entry into the healthcare system. They can refer patients to specialists in the outpatient sector and/or to a hospital if necessary. GPs also usually certify sick leave.

Employees are also automatically enrolled in Unemployment Insurance (UI), which is funded by a payroll tax shared equally between workers and employers. UI benefits are available regardless of the reason for separation, including dismissal for poor performance

<sup>&</sup>lt;sup>7</sup> Source: OECD Employment and Labour Market Statistics.

<sup>&</sup>lt;sup>8</sup> Source: Statistics Austria, own calculations.

<sup>&</sup>lt;sup>9</sup> Before being merged into one entity in 2020, the *Austrian Health Insurance Fund* was divided into nine separate provincial health insurance funds (*Gebietskrankenkassen*).

<sup>&</sup>lt;sup>10</sup> Insured individuals over the age of 18 can participate in a general health check-up program. This is usually performed by GPs and includes a medical history and a series of age- and gender-specific diagnostic and laboratory tests to detect health risks and diseases at an early stage. Men have the option of a regular prostate-specific antigen (PSA) blood test, while women over the age of 40 are entitled to a mammography screening every two years to detect breast cancer (Pruckner et al., 2020).

or misconduct. Laid-off workers are entitled to benefits immediately upon becoming unemployed, while workers who quit their jobs receive benefits only after a one-month waiting period. The minimum replacement rate is 55 percent of the daily net wage before unemployment. There are higher replacement rates for workers with dependents. The duration of UI benefits depends on the age and the labor market history of the worker. After UI benefits are exhausted, means-tested income support is available (Ahammer et al., 2023).

The public pension system covers all private-sector workers in Austria and provides early retirement, old-age, and disability pensions. Public pensions are the main source of income for retirees in Austria. The amount of the pension depends on the number of insurance months accumulated during the individual's working life and their income history (Frimmel, 2021). Compared to other countries, Austrian pensions are relatively generous. The replacement rate of 74.1 percent is well above the OECD average of 50.7 (OECD, 2023). The statutory retirement age is 65 for men and 60 for women. However, the actual retirement age is much lower. In 2022, the average retirement age was 61.6 for men and 59.7 for women (Dachverband der Sozialversicherungsträger, 2023). This gap is mainly explained by early retirement due to disability and schemes that were introduced to smooth the transition to retirement (e.g., part-time retirement schemes). Employers also influence their workers' decision to retire by offering special severance payments (i.e., golden handshakes), especially when wages for older workers are high (Frimmel et al., 2018).

### 2.2 Data Sources

We use two comprehensive administrative data sets to analyze the effect of severe health shocks of workers on their coworkers. Detailed information on daily labor market activity is available from the *Austrian Social Security Database (ASSD)*. The *ASSD* is structured as a matched worker-firm data set. The data record all employment, unemployment, and retirement spells of all private sector workers in Austria between 1972 and 2018 (Zweimüller et al., 2009). Based on these spells, we identify the main employer of each worker in a given quarter, which is important for identifying the coworkers of those workers who suffer a health shock. The data set also provides information on firm characteristics, workers' socio-demographic characteristics, and earnings.

The ASSD data can be linked to health record data from the Upper Austrian Health Insurance Fund (UAHIF). These data provide detailed information on outpatient visits, prescribed medications, and hospital visits for more than one million private sector employees and their dependents in Upper Austria from 2005 to 2018.<sup>11</sup> The data on hospital visits also include the admission diagnoses according to the International Classification of

<sup>&</sup>lt;sup>11</sup> With more than 1.5 million inhabitants (16.73 percent of the total Austrian population), Upper Austria is the third largest of Austria's nine federal states (see Statistics Austria).

Diseases (ICD-10) scheme. We use this information to identify the relevant health shocks (see Section 3.1).

#### 2.3 Outcome Variables

Labor Market Outcomes: Regarding the labor market activity of coworkers, we examine employment, earnings, and retirement. For employment, we use the ASSD to obtain the number of days of employment in a quarter. We also consider the probability of being employed in the firm where the health shock occurred. To do this, we obtain a worker's main employer in each quarter and define a binary indicator that equals one if a worker is employed in the shock firm in a given quarter and zero otherwise. The ASSD also provides information on workers' earnings. These earnings are top-coded by the maximum social security contribution base.<sup>12</sup> Because no information on hours worked is available, we can only consider daily earnings. We also analyze wage growth by calculating the difference in log daily earnings between two consecutive quarters. Retirement is measured by the number of retirement days in a given quarter. This includes all types of retirement (i.e., regular, disability and early retirement).

It has been shown that Austrian firms offer golden handshakes to influence their workers' retirement decisions (Frimmel et al., 2018). Using data from the Austrian Ministry of Finance, we also examine whether coworkers receive a golden handshake after the health shock. The data include information on special severance payments. We classify these special severance payments as golden handshakes if the amount of the severance payment exceeds the monthly earnings by a factor of 8 or more. Since the data from the Ministry of Finance are only available until 2012, we analyze this outcome only for health shocks that occurred before 2010.

*Health Outcomes:* The link to the *UAHIF* data allows us to study the effects of health shocks on coworkers' health (behavior). We use information on sick leaves to measure coworkers' absenteeism. We consider both the number of sick leave days associated with all diagnoses in a given quarter and the number of days associated with easy-to-fake diagnoses, which are associated with symptoms that are not readily visible and therefore difficult for physicians to verify.<sup>13</sup> Previous research has shown that these sick leaves are related to shirking (Ahammer, 2018), so this outcome can also be interpreted as a surrogate measure of worker effort.

With respect to health behavior, we consider the use of various health screening programs as an outcome. Specifically, we use the information on outpatient expenditures pro-

<sup>&</sup>lt;sup>12</sup> The maximum social security contribution base changes from year to year. In 2024, it is 6,060 Euros per month, which is equivalent to 202 Euros per day (see <u>oesterreich.gv.at</u>).

<sup>&</sup>lt;sup>13</sup> These diagnoses include only sick leaves related to the common cold (ICD-10 J04 and J06), low back pain (ICD-10 M54.5), and headache (ICD-10 R51).

vided by the *UAHIF* data to examine the probability of having a general health check-up, which includes several tests for cardiovascular and cancer diseases, a PSA test (only for men), or a mammography (only for women) in a given quarter. For each screening program, we define separate binary indicators that are equal to one if expenditures for that screening program are greater than zero in a given quarter, and equal to zero otherwise.

# 3 Empirical Strategy

To estimate the causal effect of a worker's severe health shock on coworkers' labor market and health outcomes, we would ideally compare coworkers in a firm who experience such a shock with identical coworkers who do not. We address the key challenge of constructing a valid counterfactual for treated coworkers by combining a matching strategy with a difference-in-difference model. We first describe the identification of health shocks and the matching strategy in detail before explaining the construction of the estimation sample and the empirical model.

## 3.1 Identification of Health Shocks and Matching

To construct an appropriate control group for those coworkers who experience a severe health shock of one of their colleagues, we follow the approach implemented by Jäger and Heining (2022). First, we identify severe health shocks of workers in the administrative health data. This allows us to divide firms and their workers into those that experience a severe health shock of a worker and those that do not. Second, we implement a combination of exact and propensity score matching to find a statistical twin for each worker-firm pair with an actual health shock in the pool of those workers employed in firms without a health shock. The treatment group comprises all worker-firm pairs where a worker actually experiences a severe health shock. The control group consists of firms where no health shock occurs, but where there is a placebo shock worker who is similar to an actual shock worker. Finally, we add information on the coworkers to construct our final estimation sample. We describe each step in more detail below.

Identification of Severe Health Shocks: In a first step, we identify severe health shocks based on the admission diagnoses recorded in the hospital data. The health shocks we consider are cancer and circulatory diseases. Specifically, we consider all types of malignant neoplasms (ICD-10 C), myocardial infarction (ICD-10 I21 and I22), heart failure (ICD-10 I50), and cerebrovascular disease (ICD-10 I60 to I66). For each individual, we keep the first severe health shock diagnosis observed in the data. We then keep only those individuals, who were employed and aged between 15 and 65 at the time of the shock, and aggregate the individual worker-level information on the severe health shocks at the firm level. To do this, we keep the first shock diagnosis observed among all workers with health shocks in a given firm. To ensure that we can accurately identify the worker responsible for the health shock, we drop firms with multiple first shocks in the same quarter.

We then combine this sample of firms with a health shock with the sample of all other firms. In order to have a consistent sample of firms, we include only firms that (i) employ more than three workers in all quarters that a firm is observed, (ii) are observed without any gaps, (iii) are observed for at least four consecutive quarters, and (iv) where all workers are insured in Upper Austria. The last restriction is necessary to ensure that firms in the donor pool do not actually experience a health shock. Since the hospital data are only observed for workers insured in Upper Austria, we can only be sure that no relevant hospital diagnoses were recorded for a worker in a given firm only if all workers in the firm are insured in Upper Austria.

*Matching Procedure:* Based on the information on whether a firm experiences a severe health shock of a worker or not, we divide the sample into a treatment group and a donor pool. The treatment group includes all worker-firm pairs that experience a worker's first severe health shock between 2006 and 2018. The donor pool includes all worker-firm pairs of firms that do not experience a severe health shock until 2018.<sup>14</sup>

For each worker-firm pair with an actual severe health shock experienced in quarter s, we select a worker-firm pair from the donor pool with similar lagged observable characteristics. We do this using a combination of exact and propensity score matching. In this way, for each worker-firm pair treated in quarter s, we construct a control group consisting of similar worker-firm pairs from firms that do not experience a severe health shock. The list of variables used in the matching procedure includes firm and worker characteristics. The firm characteristics include the number of workers, industry at the 2-digit level, the median wage, the age composition of the workforce, the age of the firm, the share of women, the share of white-collar workers, the share of blue-collar workers, and the share of non-Austrian citizens in the workforce. For the worker characteristics we choose age in five-year categories, daily earnings, and binary indicators for workers' collar, gender, citizenship, and for whether a worker is insured in Upper Austria.<sup>15</sup> All variables are measured four quarters before the health shock (i.e., in s - 4). Thus, we are matching on the levels and not on the trends of these variables, which means that the pre-trends can still be used as an appropriate validity check for the parallel trends assumption.

Throughout the matching process, which we perform separately for each shock quarter

<sup>&</sup>lt;sup>14</sup> To ensure that the firms in the donor pool are truly untreated, we use information on hospital diagnoses from 1998 onward. This allows us to remove firms from the donor pool that could have experienced a health shock at any time after 1998 but before 2006.

<sup>&</sup>lt;sup>15</sup> The variables printed in *italics* are used in the exact matching, while the other characteristics were used to compute the propensity score.

s, we ensure that each firm and worker in the donor pool is used only once as a matching pair. In cases where there are multiple closest matches in the donor pool, one of them is selected randomly. Using this procedure, we can find a match for 45 percent of all worker-firm pairs with a severe health shock. Table A.1 in the Appendix compares firm and worker characteristics of matched and unmatched worker-firm pairs. Matched firms are on average smaller than unmatched firms.<sup>16</sup> They also employ more women and have a smaller share of blue collar workers and foreigners in their workforce. Average daily wages are 7.5 Euros lower in matched firms than in non-matched firms. Consistent with these firm-level differences, there are more women and fewer non-Austrian citizens among the matched workers, and their wages are lower than those of non-matched workers. Matched workers are also more likely than non-matched workers to be employed in the shock firm four quarters before the health shock. They also have lower healthcare expenditures, but these differences are not statistically significant. Panel (a) of Figure 1 shows that matching success is relatively stable over time. The propensity scores in the treatment and control group are very similar, indicating that our approach leads to close matches (Panels (b) and (c)).

Construction of the Estimation Sample: So far, we have focused mainly on those workers who experience a severe health shock, identifying a statistical twin for each of them in the data. However, we are ultimately interested in the outcomes of the coworkers of these shock workers and their matched controls. Therefore, in a final step, we add information on all workers employed at the firm where a (placebo) health shock occurs in quarter s, excluding of course the observations of the workers responsible for the health shock. In the end, we compare the coworkers of workers who actually experience a severe health shock with coworkers who have a colleague without a health shock but who is similar to an actual shock worker on key observable characteristics and who is employed in a firm without a health shock that has similar characteristics to a firm where an actual health shock occurred.

For our main estimation sample, we construct a panel at the coworker-quarter level that tracks coworkers for up to 12 quarters before and 12 quarters after the (placebo) health shock. We impose several sample restrictions. First, we restrict the sample to firms that employ no more than 50 workers four quarters before the health shock. This should increase the likelihood that coworkers are actually aware of their colleague's health shock. Since we do not have information on the detailed organization of firms and workers' occupations, we cannot be sure that workers actually know each other. However, this is more likely to be the case in smaller firms than in larger ones. This restriction also allows

<sup>&</sup>lt;sup>16</sup> One explanation for the difference in firm size between matched and unmatched firms is that we match on the exact firm size. Since the probability of finding an untreated firm with the exact same firm size as a treated firm is higher for smaller firms than for larger firms, the sample of matched firms tends to include more small firms.

us to clearly identify the first severe health shock in a firm, as larger firms are likely to experience multiple health shocks of workers.

Second, we restrict the sample to include only shock workers and coworker who have been employed at the shock firm for at least four quarters prior to the health shock (including the quarter of the health shock). Again, this should increase the likelihood that workers actually know each other and are aware of a colleague's severe health shock.

Third, because firms and coworkers may experience a second health shock of a worker during the sample period, we only include observations up to the quarter of the second shock at the firm and/or coworker level. In our sample, about 18 percent of treated firms experience a second health shock of a worker, with an average duration of 12 quarters between the two events. Only 0.8 percent of coworkers experience a second health shock of one of their colleagues.<sup>17</sup>

Lastly, our main estimation sample focuses on coworkers who are older than 50 years at the time of the (placebo) health shock. We believe that this age group is most relevant for the outcomes considered in this paper. The labor market outcomes include the number of retirement days in a given quarter. Since retirement decisions are usually made at the end of a career, this outcome is only relevant for older workers. In addition, health screening programs are tend to target older individuals. For PSA tests, men over the age of 45 are advised to undergo them regularly.<sup>18</sup> A mammography should be done every two years from the age of 40 onward. Women between the ages of 45 and 74 receive an invitation letter from the health insurance to remind them to undergo the procedure.<sup>19</sup> For completeness, we also show the results for younger coworkers in the Appendix.<sup>20</sup>

Descriptive Statistics: Table 1 provides descriptive statistics comparing the characteristics of coworkers in the treatment and control group before the (placebo) health shock. Coworkers are about 55 years old at the time of the (placebo) health shock, work on average 89 days per quarter, and earn approximately 72 Euros per day. The probability of being employed in the shock firm before the health shock is 0.92. 51.6 percent of all coworkers are women and 6.3 percent are non-Austrian citizens. Overall, the differences between coworkers in the treatment and control group are small and not statistically significant. One exception is medication expenditure. Coworkers in the control group spend about 9 Euros more per quarter than those in the treatment group. The average firm in the sample employs 16.7 workers before the health shock. Firms in the treatment group have a lower share of women and blue-collar workers than firms in the control group,

<sup>&</sup>lt;sup>17</sup> Note also that a coworker may appear as a coworker for multiple (placebo) shock workers at different times. In our sample, however, 99.58 percent of coworkers appear only once.

<sup>&</sup>lt;sup>18</sup> See Österreichische Krebshilfe for details.

<sup>&</sup>lt;sup>19</sup> See Österreichische Krebshilfe for details.

<sup>&</sup>lt;sup>20</sup> Note that we apply the age restriction only to coworkers. Shock workers can be younger than 50 at the time of the (placebo) health shock.

while average wages are slightly higher in treatment firms than in control firms. None of the differences are statistically significant.

In Table A.2 in the Appendix, we report descriptive statistics for the characteristics of the shock workers included in the main estimation sample. The average shock worker is about 50 years old at the time of the (placebo) health shock. Like their coworkers, they are highly attached to the labor market and the shock firm. Approximately half of the shock workers are women and 2.6 percent of them are non-Austrian citizens. Compared to the coworkers, the shock workers earn slightly less. Again, the differences between the treatment and control group group are small and not statistically significant. However, as would be expected, workers who actually experience a health shock have higher healthcare expenditures than workers with a placebo shock.

Figure 2 shows the distribution of the different types of severe health shocks in the main estimation sample. More than two-thirds of all health shocks are associated with malignant neoplasms. As Table A.3 in the Appendix shows, the most common types of cancer are malignant neoplasms of the breast (ICD-10 C50), skin (ICD-10 C43 and C44), male genital organs (ICD-10 C60 to C62), and digestive organs (ICD-10 C15 to C26). 17.95 percent of health shocks are related to cerebrovascular diseases, while myocardial infarction accounts for 11.45 percent. 2.78 percent suffer from heart failure.

#### 3.2 Estimation Strategy

To estimate the effect of a severe health shock on the labor market and health outcomes of coworkers, we estimate the following two-way fixed effects (TWFE) model:

$$y_{iq} = \alpha_i + \beta \cdot \mathbb{1} \left( q \ge 0 \right)_q + \delta \cdot \left[ \mathbb{1} \left( q \ge 0 \right)_q \times HS_i \right] + \gamma \boldsymbol{X}_{iq} + \varepsilon_{iq}$$
(1)

where  $y_{iq}$  is the outcome of interest of coworker *i* observed in the relative quarter *q*. The relative quarter is defined as the difference between the calendar quarter *t* and the quarter of the (placebo) health shock *s* (i.e., t - s).  $\alpha_i$  are coworker fixed effects,  $\mathbb{1} (q \ge 0)_q$  is a binary indicator that is equal to one for the post-treatment period and zero for the pre-treatment period, and  $HS_i$  is a binary indicator equal to one for those coworkers who actually experience the health shock of one of their colleagues and zero for those coworkers with a placebo health shock.  $\delta$ , the coefficient of the interaction term  $\mathbb{1} (q \ge 0)_q \times HS_i$ , is the coefficient of interest. It measures the average effect of experiencing an actual severe health shock on the outcome of interest as compared to no shock occurring.  $X_{iq}$  includes calendar year  $\times$  birth year fixed effects to flexibly control for age and general trends in the outcome variable. Since all coworkers in a shock firm are either treated or not treated, standard errors are clustered at the shock firm level.

Interpreting  $\delta$  as the causal effect of a severe health shock on coworkers requires that the outcomes of coworkers in the treated group would have evolved in the same way as in the control group in the absence of the health shock. While the parallel trends assumption is not directly testable, one can check whether coworkers' outcomes are already on different trends before the health shock. If this is not the case, it is likely that the outcomes would have actually evolved in the same way in the absence of the health shock. We show that coworkers' outcomes are not on diverging trends before the (placebo) health shock by estimating the following dynamic TWFE model:

$$y_{iq} = \alpha_i + \sum_{k=-12, \ k\neq-4}^{11} \beta_k \cdot \mathbb{1} (q=k)_q + \sum_{k=-12, \ k\neq-4}^{11} \delta_k \cdot \left[ \mathbb{1} (q=k)_q \times HS_i \right] + \gamma X_{iq} + \varepsilon_{iq}$$
(2)

where  $\mathbb{1} (q = k)_q$  is a series of binary indicators for the different relative time periods. The coefficients  $\delta_k$  denote the treatment effects of interest relative to the reference period of four quarters before the (placebo) health shock. Setting the reference period to q = -4 allows us to estimate treatment effects for the last quarters before the health shock and to test for the existence of anticipation effects. All other variables are defined as in Equation (2).

In our empirical setting, not all coworkers experience a severe health shock of one of their colleagues at the same time. Such variation in treatment timing may be problematic if treatment effects vary over time (Goodman-Bacon, 2021). To address this potential concern, we also provide estimates for the model specified in Equation (2) based on the method developed by Sun and Abraham (2021) in the Appendix.

## 4 Results

We first focus on those workers that actually suffer a severe health shock. This initial analysis shows that the health shocks considered in this study are indeed a relevant and significant event in the life of these workers in terms of their health and labor market activity. This analysis also helps to understand the extent of the need to replace a shocked worker within a firm. Knowing about the magnitude and relevance of the treatment, we then examine the impact of the severe health shocks on coworkers' labor market outcomes and health behavior.

#### 4.1 Effect of Health Shocks on Shock Workers

We consider different types of healthcare expenditures to verify that our empirical strategy indeed identifies severe and unexpected health shocks that have a long-lasting negative impact on the health of those who suffer them. The first column of Figure 3 plots average inpatient, outpatient, and medication expenditures for each relative period separately for treatment and control shock workers. The second column shows the estimates from the dynamic TWFE model specified in Equation (2). Prior to the health shock, there are no differences in expenditures between workers in the treatment and control group, suggesting that the health shocks are unexpected. However, in the quarter of the health shock, expenditures in all three categories increase substantially for treated workers. The average health shock increases inpatient expenditures by about 9,700 Euros, outpatient expenditures by about 78 Euros, and medication expenditures by about 297 Euros in the quarter of the shock. Although the estimated coefficients decrease in subsequent periods, persistent differences in health status between treated and control workers remain. Even 12 quarters after the health shock, inpatient expenditures of treated workers are about 850 Euros higher than those of control group workers compared to four quarters before the health shock. Outpatient expenditures are about 40 Euros higher and medication expenditures for PSA tests, mammography, dermatology, and radiology (see Figure A.1 in the Appendix) increase significantly in the quarter before and the quarter of the health shock. This suggests that standard preventive screening tools are responsible for the detection of the health shock diagnoses.

Another indicator for the severity of the health shock is the increase in mortality. As Panel (a) of Figure A.2 in the Appendix shows, treated workers have an 11 percent chance of dying within 12 quarters after the shock, while it is virtually zero for the control group. Panel (b) shows that the type of the health shock has important implications for the increase in mortality. While the immediate increase in mortality associated with cardiovascular diseases is higher than that associated with cancer, the long-run increase is much larger for those with a cancer diagnosis as their first health shock. Also, those aged below 50 at the the time of the health shock have a lower probability of dying within 12 quarters after the shock (Panel (c)).

The severe health consequences for those who suffer the shocks considered in this paper also have a significant impact on their labor market participation. As Panels (a) and (b) of Figure 4 show, those workers who suffer a health shock considerably reduce the number of days of employment. The largest decrease of about 30 days is observed two quarters after the health shock. This corresponds to a 33 percent decrease compared to the pre-shock sample average. Although this gap narrows over time, 12 quarters after the health shock, treated workers spend 10.87 fewer days in employment than those without a health shock. Most importantly, this reduction in employment is a relevant factor for the firm in which shock workers are employed at the time of the shock. Panels (c) and (d) of Figure 4 show that workers with a health shock are 25 percentage points less likely to work at the shock firm two quarters after the health shock. Although some workers appear to return after some time, there is still a gap of 7.3 percentage points relative to the control group after 12 quarters. Thus, these health shocks do indeed represent a significant loss of human capital that must be replaced by new hires or compensated by current coworkers. There is some heterogeneity in the employment response with respect

to the shock type and the age at the time of shock. While the immediate reduction in employment is larger for cancer patients, they face a smaller long-term effect than those suffering from a cardiovascular disease (Panel (e)). Also, workers under 50 tend to return to the labor market more often in the long run than older workers, although the immediate employment responses are similar for both groups (Panel (f)).

Even if workers with a health shock remain in the labor market, there is a substantial reduction in their actual labor supply. As Panel (a) of Figure A.3 in the Appendix shows, the health shock leads to a persistent increase in absenteeism. The average increase in sick leave days is about 36 days in the quarter of the health shock, which is 14 times higher than the pre-shock average. Even 12 quarters after the health shock, treated workers have 2.1 more sick leave days than workers in the control group. Importantly, Panel (b) shows that this increase in sick leave days is not driven by sick leaves associated with easy-to-fake diagnoses, such as back pain. This implies that the increase in absenteeism is not due to shirking, but is actually caused by a reduction in work capacity.

Panel (c) of Figure A.3 in the Appendix shows that the reduction in labor supply of those who suffer the severe health shocks is also reflected in an increase in retirement days. Five quarters after the health shock, workers in the treatment group spend about 4 to 6 days more per quarter in retirement than workers in the control group. The average control group worker spends 3.8 days in retirement in the fourth quarter after the health shock, implying that workers with a health shock spend more than twice as many days in retirement.<sup>21</sup> The increase in the total number of retirement days is driven by an increase in disability retirement (Panel (d)).

Overall, the health shocks are a significant event not only for the affected workers, but also for the firms in which these workers are employed at the time of the shock. Firms are affected both by an increased probability of workers leaving the firm and dropping out of the labor market and by workers taking more sick leaves. Thus, firms need to compensate for this reduction in their labor force. Because of the partial nature of the reduction in employment, the compensation is more likely to come from existing coworkers than from new hires.

#### 4.2 Effect of Health Shocks on Coworkers

Labor Market Outcomes: The previous section shows that the health shocks studied in this paper have substantial and long-lasting negative effects on the labor market and health outcomes of those who suffer them. We now turn to the effects of these health

<sup>&</sup>lt;sup>21</sup> To construct our estimation sample, we require that workers be employed at the shock firm in the quarter of the shock and the three quarters preceding the shock. Because of this requirement, the number of retirement days prior to the health shock is close to zero. For all outcomes related to retirement, we report the sample average of the control group in the post-period or in relative quarter 4 as the mean of the outcome in all tables and figures, as we believe this is a more appropriate value for gauging the size of the estimated coefficients.

shocks on coworkers by estimating Equation (1) for the labor market and health outcomes of coworkers. Table 2 reports the average treatment effects for coworkers' employment, earnings, and retirement. Compared to coworkers in the control group, those who actually experience a severe health shock of one of their colleagues have on average 1.07 more days of employment per quarter after the health shock. This corresponds to an increase of 1.20 percent relative to the average number of quarterly employment days before the treatment. The increase in employment days is also reflected in the probability of being employed in the firm where the health shock occurred. The health shock increases coworkers' probability of working at the shock firm by 2.32 percentage points or 2.52 percent. Both coefficients are statistically significant at the 10 percent level (p-values 0.083 and 0.095). Consistent with the finding that treated coworkers are more attached to the labor market, Column (3) shows that the coworkers with a health shock spend, on average, 1.05 fewer days per quarter in retirement than coworkers in the control group. The point estimate is statistically significant at the 5 percent level and is of a similar absolute size as the increase in employment days.

Columns (4) and (5) of Table 2 report the coefficients for coworkers' daily earnings and earnings growth. The estimated coefficients are not statistically significant for either outcome. Although this suggests that firms do not compensate coworkers for staying longer at the shock firm with higher wages, there may be other forms of remuneration that firms could choose to reward their employees. One such remuneration could be special severance payments (i.e., golden handshakes) when coworkers leave the firm. We analyze the probability of receiving a golden handshake from the shock firm after the health shock by estimating a cross-sectional regression including a binary treatment indicator, a set of coworker and shock worker characteristics, and shock year fixed effects.<sup>22</sup> Table A.4 in the Appendix reports the results for all coworkers employed at the shock firm in the quarter of the health shock in Columns (1) and (2), and for the subsample of coworkers whose last employment spell before retirement was at the shock firm in Columns (3) and (4). The estimated coefficients suggest that coworkers who experience a severe health shock of one of their colleagues are more likely to receive a golden handshake after the health shock occurred. This is also true for coworkers who leave the shock firm for retirement. This implies that rather than adjusting base pay to compensate coworkers for staying longer at the shock firm, firms reward treated coworkers through other forms of remuneration.

*Health Outcomes:* Coworkers' labor market outcomes may not be the only dimension that can be affected by a health shock. The experience of a health shock may induce coworkers to reconsider their own health situation and, at least temporarily, increase

 $<sup>^{22}</sup>$  Information on special severance payments is only available until 2012. We therefore restrict the estimation sample for this analysis to all health shocks prior to 2010. We classify a special severance pay as a golden handshake if the amount of the special severance pay exceeds the monthly earnings by a factor of 8 or more.

their preventive health behavior. Therefore, we also analyze the effect of the health shock on coworkers' health (behavior). Table 3 reports the estimated average effects for sick leave days and participation in different forms of health screening programs. Column (1) shows that there is no statistically significant effect on the number of sick leave days. This is also true for sick leave days related to easy-to-fake diagnoses, such as the common cold, lower back pain, and headache. Since sick leaves resulting from these diagnoses have been shown to be related to shirking (Ahammer, 2018), the absences of a statistically significant effect suggests that the health shock does not lead to a change in coworkers' effort.

With respect to coworkers' health behavior, we find some evidence of an increase in the participation in health screening programs.<sup>23</sup> Although the estimated coefficients for the probability of participating in the general health check and undergoing a PSA test in Columns (3) and (4) of Table 3 are positive and non-negligible in magnitude, they are imprecisely estimated. For mammographies, we find that female coworkers with a health shock are 0.64 percentage points (9.73 percent) more likely to have the procedure compared to coworkers in the control group. The estimated coefficient is statistically significant at the 10 percent level (p-value 0.060).

*Discussion:* Our main results suggest that a severe health shock of a worker generally increases the employment of their coworkers. In particular, it increases the probability of working at the firm where the shock occurred. We also find a small but statistically significant delay in retirement. This implies that firms compensate for the loss of the shock worker's working capacity with with incumbents rather than with external hires. While this does not lead to an increase in wages, firms still reward coworkers with special severance payments when they leave the firm for retirement. Importantly, the health shock does not increase coworker absenteeism, and we find suggestive evidence that it does not affect effort. However, witnessing the severe health shock of a coworker does induce changes in health behavior. We find an increase in the probability of participating in health screening programs, especially for mammographies.

Compared to the estimates in Jäger and Heining (2022), our point estimates for the probability of being employed at the shock firm are considerably larger. While Jäger and Heining (2022) find that the death of a worker increases coworkers' probability of being employed at the firm where the death occurred by 0.3 percentage points, or 0.39 percent, we find an increase of 2.32 percentage points, or 2.52 percent.<sup>24</sup> We also find no statistically significant effect on daily wages and wage growth. However, we do find that treated workers are more likely to receive a golden handshake when they leave the

<sup>&</sup>lt;sup>23</sup> Note that PSA tests are analyzed only for men and mammographies are analyzed only for women.

<sup>&</sup>lt;sup>24</sup> It is important to note that our main results refer only to coworkers aged over 50 at the time of the health shock, which may limit the comparability of the estimated coefficients. Also, the partial lack of statistical precision does not allow a strong statement about the comparison of effect sizes.

shock firm for retirement. This implies that there is still an earnings gain associated with the health shock, but that this does not seem to be reflected in base pay as in Jäger and Heining (2022).

So far we have only considered coworkers who were older than 50 at the time of the health shock. In the Appendix, we also provide estimates for the same set of outcome variables for younger coworkers. We find that the health shock has, if anything, only a muted effect on young coworkers' labor market outcomes (see Table A.5 in the Appendix). While the number of employment days increases by 0.72 days (p-value 0.079), there is no statistically significant change in the probability of being employed by the shock firm. The number of retirement days increases by 0.12 days. Given that retirement is hardly relevant for this age group, we are cautious about the causal interpretation of this coefficient. As for older coworkers, there is no statistically significant effects for sick leave days and participation in health screening programs (see Table A.6 in the Appendix).

#### 4.3 Effect Heterogeneity

Our main results suggest that coworkers compensate for the firm's loss of labor supply associated with the severe health shock of one of their colleagues by being more likely to be employed at the shock firm. In general, the health shock increases employment and reduces retirement. It also affects the health behavior of female coworkers by increasing the probability of having a mammography. As a next step we shed more light on potential mechanisms and treatment heterogeneity of our effects.<sup>25</sup> First, we consider the skill level of the shock worker. Productivity losses of higher skilled workers are expected to be more "damaging" to firms and therefore open a larger need for compensation, especially for equally experienced workers. Second, because several papers show that the similarity between coworkers is relevant for behavioral responses (Fadlon and Nielsen, 2019; Pruckner et al., 2020), we estimate the effects separately by the age and gender of the shock worker and the coworker. Finally, because more stress-related cardiovascular health shocks may trigger coworkers differently than cancer diagnoses, we also consider treatment hetero-geneity by the type of the health shock.

Shock Workers' Skill: Compensating for the reduced labor supply with an external hire may be more difficult for firms when the shock worker is highly skilled. This implies that the increase in the probability of incumbent workers being employed by the shock firm should be larger when a more skilled worker suffers a severe health shock. Since worker skill is not directly observable in our data, we use two surrogate measures. First, since

<sup>&</sup>lt;sup>25</sup> As the estimated coefficients for some outcomes are not statistically significant at any conventional level, we present the results of the heterogeneity analysis only for those labor market and health outcomes for which the coefficients are statistically significant at least at the 10 percent level in the full sample.

wages should be correlated with workers' skills, we split the sample by the shock worker's position in the shock firm's wage distribution. Table 4 reports the results separately for coworkers of shock workers who are below or above the 75<sup>th</sup> percentile of the within-firm wage distribution. As expected, the labor market effects are driven by high-wage (i.e., high-skilled) shock workers. For their coworkers, the health shock increases the probability of being employed at the shock firm by 5.86 percentage points (6.34 percent). This increase in labor supply to the shock firm is reflected in an overall increase in employment days and a decrease in retirement days. For the coworkers of less skilled shock workers, the point estimates have the same sign as those for the coworkers of more skilled shock workers, but they are considerably smaller in size and also not statistically significant. The effect on the probability of having a mammography is positive and non-negligible in magnitude for both groups of coworkers, but is only statistically significant at the 10 percent level for coworkers of less skilled shock workers (p-value 0.071).

As a second measure of worker skill, we use estimated Abowd et al. (1999) (AKM) worker wage fixed effects. A higher AKM worker fixed effect should reflect higher worker productivity, which should be positively correlated with the skill level. Separate results for workers with AKM worker fixed effects above and below the median are reported in Table 5. Again, the estimated coefficients on labor market outcomes are larger for coworkers of productive shock workers than those for coworkers of less productive shock workers. For the latter group, none of the estimated coefficients are statistically significant at any conventional level. The effect on mammographies is of similar magnitude for both groups, but neither coefficient is statistically significant.

Panel (f) of Figure 4 shows that the shock workers' employment response to the Age: severe health shock differs by the age of the shock worker at the time of the shock. We therefore estimate Equation (1) separately for coworkers whose colleague with the health shock is below or above 50 at the time of the shock. Table A.7 in the Appendix reports the results, which show that health shocks to younger workers increase their coworkers' employment by 1.91 days, increase the probability of being employed at the shock firm by 3.32 percentage points (p-value 0.083), and reduce retirement by 1.74 days. For the coworkers of older shock workers, the point estimates are smaller in size and not statistically significant at any conventional level. The effect for mammographies, on the other hand, is driven by shock workers over the age of 50. In this group, coworkers with a health shock are 0.95 percentage points (15.4 percent) more likely to have a mammography than coworkers in the control group. The point estimate for coworkers of younger shock workers is smaller and not statistically significant. Overall, these results suggest that older coworkers compensate for the reduced work capacity of younger shock workers and that the similarity between shock workers and coworkers is important for the change in health behavior.

That older coworkers play an important role in cushioning the adverse effects of the severe health shock on the available labor force is further supported by the results in Table A.8 in the Appendix. It reports separate estimates for coworkers younger than 55 (i.e., aged 50 to 54) and those older than 55 at the time of the health shock. The point estimates, although some are imprecisely estimated, suggest that the increase in labor supply is concentrated among coworkers aged 55 and older. For mammographies, the estimated coefficients are non-negligible in size, although only the estimate for coworkers younger than 55 is statistically significant at the 10 percent level (p-value 0.10).

Gender: Table A.9 in the Appendix reports the estimated coefficients separately for women and men.<sup>26</sup> We find that the increase in employment days and employment at the shock firm is concentrated among men, while the reduction in retirement days is statistically significant only for women. This implies that the increase in employment for men is not reflected in the decrease in their retirement, and the decrease in retirement for women is not reflected in their employment. Two factors could explain this result. First, the statutory retirement age for women is lower for women than for men (60 vs. 65), which means that, at any age above 50, retirement is a more relevant substitute for employment for women than for men. This is reflected in the outcome means reported in Table A.9 in the Appendix, as women in the control group spend on average 4.2 more days in retirement in the post-treatment period than men in the control group. Thus, the increase in men's employment could be reflected in a decrease in alternative labor market spells other than retirement, such as unemployment. Second, the signs of the point estimates for women's employment and men's retirement are consistent with the estimates for women's retirement and men's employment, but they are imprecisely estimated. For example, the lower bound of the 95 percent confidence interval of the point estimate for men's retirement would suggest a reduction of 1.20 days. This is similar in magnitude to the increase in employment of 1.67 days. Similarly, the upper bound for women's employment days is an increase of 2.07 days, which is similar in absolute magnitude to the reduction in retirement of 1.42 days.

To take the gender of the shock worker into account, we evaluate the effects separately according to whether the shock worker and the coworkers have the same gender or not in Table A.10 in the Appendix. While the increase in employment at the shock firm appears to be driven by cases where the coworkers are of the same gender as the shock worker (p-value 0.075), the coefficients are not estimated with sufficient precision to conclude that the labor market effects are indeed different between the two groups. Consistent with the argument that the changes in health behavior are more likely to occur when coworkers are more similar to the shock worker, we find that the probability of having a

<sup>&</sup>lt;sup>26</sup> Note that mammographies are analyzed only for women. This outcome is therefore omitted from Table A.9 in the Appendix.

mammography increases by 0.91 percentage points (13.39 percent) when the shock worker is also a woman. Women do not change their participation in mammographies when the health shock affects a man.

Shock Type: Panel (e) of Figure 4 shows that the employment responses of shock workers differ between severe health shocks due to cancer and those due to cardiovascular disease, especially in the long run. We therefore also split the sample of coworkers along this dimension and report the results in Table A.11 in the Appendix. The estimates for the labor market outcomes do not suggest a concentration of the effects among a particular group. While the increase in employment days and the reduction in retirement days are larger for cancer-related health shocks, the increase in the probability of being employed in the shock firm is larger for cardiovascular diseases. However, most of the coefficients and the differences between them are not statistically significant. As expected from the fact that mammographies are used to detect breast cancer, we find that the increase in the probability of having one is clearly driven by health shocks due to cancer. Female coworkers who experience the cancer diagnosis of a colleague are 0.88 percentage points (13.49 percent) more likely to have a mammography than those in the control group. For health shocks due to by cardiovascular diseases, the point estimate is negative but not statistically significant.

Summary: The heterogeneity analysis shows that the increase in coworkers' employment is particularly strong when a severe health shock affects a highly skilled worker. We find larger point estimates in cases where the shock worker is in the upper part of the wage distribution and for shock workers with an above-median AKM worker fixed effect. This is consistent with the hypothesis that it is more difficult for firms to compensate for the loss of labor supply caused by the health shock with an external hire if the shock worker is more skilled. Instead, incumbent workers step in. In particular, we find that older coworkers fill the gap when the shock worker is relatively young. In addition, the employment effects seem to be concentrated among male coworkers and in cases where the shock worker and coworker are of the same gender. It is important to note, however, that the coefficients for each subgroup are sometimes imprecisely estimated, which is why they provide at best suggestive evidence.

For the effects on health behavior, our results suggest that the similarity between the shock worker and the coworkers is important. We find larger increases in the probability of having a mammography when both groups of workers are in the same age group and are of the same gender. As expected from the intention of the procedure, cancer-related health shocks are mainly responsible for the overall increase. Again, the lack of statistical precision prevents us from drawing stronger conclusions.

#### 4.4 Robustness

Dynamic TWFE Estimates: So far, we have only considered the average post-treatment effect of a severe health shock on coworkers' outcomes estimated from Equation (1). This specification does not allow us to analyze the development of the outcome variables in the pre-treatment period, which is commonly used as a validity check for the parallel trends assumption. It is also not possible to determine whether the estimated treatment effects are constant or change over time. Therefore, to provide a more complete picture of the main results, we also estimate the dynamic TWFE model specified in Equation (2).

Figure A.4 in the Appendix shows the results for coworkers' labor market outcomes. Overall, the estimated coefficients in the period before the (placebo) health shock are not statistically significant. The outcomes of coworkers in the treatment and control group do not follow different trends before the treatment, suggesting that this is likely to have been the case in the absence of the treatment. Regarding the post-treatment period, we find positive point estimates for employment days (Panel (a)) and the probability of working in the shock firm (Panel (b)) in all quarters after the health shock. Although this generally confirms the effects documented in Table 2, the estimated coefficients are imprecisely estimated and not statistically significant. Consistent with the positive point estimates for employment, coworkers in the treatment group spend fewer days in retirement in all quarters after the health shock (Panel (c)). The point estimates appear to increase slightly over time and are only statistically significant at the 5 percent level in later quarters. The dynamic point estimates for log daily wages (Panel (d)) and wage growth (Panel (e)) are close to zero and statistically insignificant throughout the entire post-treatment period.

The results for coworkers' health outcomes are shown in Figure A.5 in the Appendix. Again, the assumption of parallel trends is credibly fulfilled for these outcomes. As for the average effects in Table 3, the dynamic estimates show no statistically significant effects for all sick leaves (Panel (a)) and easy-to-fake sick leaves (Panel (b)). While not reflected in the average effect for the probability of participating in the general health screening, positive and statistically significant point estimates emerge in the quarter before and the quarter of the (placebo) health shock (Panel (c)). This suggests that the increased participation of shock workers in health screening programs prior to the health shock induces their coworkers to also participate in the general health check-up. The probability of having a PSA test does not increase as a result of the health shock. If anything, some of the statistically significant point estimates go in the opposite direction (Panel (d)). The point estimates for the probability of having a mammography fluctuate around zero and are not statistically significant (Panel (e)).

Not all coworkers in our estimation sample experience the severe health shock of one of their colleagues at the same time. In case of treatment effect heterogeneity, this variation in treatment timing may lead to biased estimates (Goodman-Bacon, 2021). To address

this concern, we re-estimate Equation (2) for all our main outcomes using the method developed by Sun and Abraham (2021). Figures A.6 and A.7 in the Appendix show the results. There are no quantitatively significant differences in the estimated coefficients between our main estimates and the Sun and Abraham (2021) coefficients.

*Other Robustness Checks:* The main results are robust to (i) restricting the sample to coworkers where the shock worker leaves the firm after the health shock, (ii) weighting the regressions, and (iii) using an estimation sample where the constraint on employment in the shock firm before the health shock is relaxed.

First, we show that the estimated effects are driven by those cases where the shock worker leaves the firm after the health shock. If our finding that coworkers compensate for the reduction in working capacity by being more likely to stay at the shock firm, the effects should be driven by those cases where the loss in labor supply is even greater (i.e., when the shock worker leaves the firm). We therefore estimate Equation (1) separately for coworkers where the shock worker leaves the firm within one quarter after the health shock and for coworkers where the shock worker stays at the shock firm. Table A.12 in the Appendix shows the results for the labor market outcomes. As expected, the employment responses are driven by cases where the shock worker leaves the firm. Coworkers are 4.41 percentage points (4.83 percent) more likely to be employed in the shock firm after the health shock if the shock worker leaves the firm, while we find no statistically significant effects when the shock worker remains in the firm. For both groups, we find no statistically significant effects on wages and wage growth.<sup>27</sup>

Second, our main results are robust to weighting the regression in Equation (1) by the inverse of the size of the shock firm in the quarter of the shock. This ensures that each shock receives the same weight.<sup>28</sup> Table A.14 in the Appendix shows the results of the weighted regressions. We find employment effects of a similar size and level of statistical significance to our main estimates.

Finally, we find qualitatively similar results when we relax the employment restriction imposed in the construction of the main estimation sample (see Section 3.1). Shock workers and coworkers in our main estimation sample must be employed in the firm where the health shock occurs for at least four quarters prior to the health shock (including the quarter of the health shock). In Table A.15 in the Appendix, we relax this constraint by reducing the employment requirement to two quarters (i.e., the quarter before and the quarter of the health shock) and find positive point estimates for the number of employment days and the probability of being employed in the shock firm, although

<sup>&</sup>lt;sup>27</sup> The coefficients for the health outcomes are reported in Table A.13 in the Appendix. However, none of the estimated coefficients are statistically significant.

<sup>&</sup>lt;sup>28</sup> As larger firms contribute more observations to our estimation sample, shocks to larger firms are implicitly given more weight than shocks to smaller firms when estimating an unweighted version of Equation (1)

the latter is somewhat smaller in size than the main estimate. In addition, coworkers in the treatment group spend fewer days in retirement and are more likely to have a mammography as a result of the health shock.

# 5 Conclusion

In this study, we provide new evidence on the spillover effects of health shocks outside family networks by extending the setting to coworkers and firms. First, we show that these health shocks are severe for those who suffer them, leading to substantial increases in healthcare expenditures, absenteeism, and mortality, as well as to significant and persistent reductions in labor supply. Combining a matching approach to construct a control group and a difference-in-difference framework, we then analyze how a worker's severe health shock affects the labor market outcomes and health behavior of their older coworkers. We find a significant impact on both dimensions. Coworkers are about 2.3 percentage points more likely to be employed in the shock firm and significantly delay retirement. Although we find no evidence for changes in daily earnings or wage profiles, firms compensate their loyal employees with special severance payments when they leave the firm. Treatment heterogeneity analysis shows that the effects on employment and retirement are larger when the health shock affects a high-skilled worker and when the shock worker leaves the firm after the health shock. In terms of health behavior, we find that female coworkers in the treatment are about 9.73 percent more likely to have a mammography after a cancer-related health shock. However, we find no statistically significant effects on participation in general health check-ups and PSA tests, or on coworker absenteeism.

Our results are consistent with the hypothesis that older coworkers step in to compensate for the (partial) loss of labor supply to the firm associated with the severe health shock. For example, we find larger employment effects when the shock worker is highly skilled. As these workers are more difficult to replace and require experienced substitutes, firms tend to rely more on incumbent workers rather than hiring external workers. As a monetary compensation, incumbent workers receive higher severance payments when they leave the firm. The findings that older coworkers expand their labor supply and adjust their preventive health behavior only marginally in the short run after experiencing a health shock in the firm speak against the alternative hypothesis that the health shocks affect coworkers through an increase in workplace stress, as this should induce coworkers to leave the shock firm.

There are important limitations to this study that somewhat limit the external validity of our findings. First, we only focus on small firms with less than 50 workers four quarters before the (placebo) health shock. These firms are often family-run and may come with better working conditions and closer relationships between coworkers. In contrast, health shocks in larger firms may elicit different reactions from coworkers, with potentially ambiguous results, as health shocks in larger firms may be less perceived or differently attributed to workplace stress. Second, we only consider deteriorations in workers' physical health, and not their mental health, as health shocks. Therefore, our findings cannot necessarily be generalized to all types of health shocks. As mental health is becoming increasingly important, especially in the workplace, more evidence is needed to provide a more complete picture of the spillover effects of health shocks in the workplace.

# References

- Abowd, John M., Francis Kramarz and David N. Margolis (1999), 'High Wage Workers and High Wage Firms', *Econometrica* 67(2), 251–333.
- Ahammer, Alexander (2018), 'Physicians, Sick Leave Certificates, and Patients' Subsequent Employment Outcomes', *Health Economics* 27, 927–936.
- Ahammer, Alexander, Gerald J. Pruckner and Flora Stiftinger (2024), The Labor and Health Economics of Breast Cancer, Working Paper No. 2409, Department of Economics, Johannes Kepler University Linz, Linz.
- Ahammer, Alexander, Matthias Fahn and Flora Stiftinger (2023), Outside Options and Worker Motivation, Working Paper No. 10581, CESifo, Munich.
- Ahammer, Alexander, Rene Wiesinger and Katrin Zocher (2021), The Austrian Healthcare System: Changes and Challenges, in B. H.Baltagi and F.Moscone, eds, 'The Sustainability of Healthcare Systems in Europe', Contributions to Economic Analysis Vol. 295, Emerald Publishing Limited, Bingley, UK, pp. 153–166.
- Angelini, Viola and Joan Costa-Font (2023), 'Health and Wellbeing Spillovers of a Partner's Cancer Diagnosis', Journal of Economic Behavior and Organization 212, 422–437.
- Åslund, Olof and Peter Frederiksson (2009), 'Peer Effects in Welfare Dependence. Quasi-Experimental Evidence', Journal of Human Resources 44(3), 798–825.
- Azoulay, Pierre, Joshua S. Graff Zivin and Jialan Wang (2010), 'Superstar Extinction', Quarterly Journal of Economics 125(2), 549–589.
- Bandiera, Oriana, Iwan Barankay and Imran Rasul (2010), 'Social Incentives in the Workplace', *Review of Economic Studies* 77, 417–458.
- Becker, Sascha O. and Hans K. Hvide (2022), 'Entrepreneur Death and Startup Performance', *Review of Finance* **26**(1), 163–185.
- Bertheau, Antoine, Pierre Cahuc, Simon Jäger and Rune Vejlin (2022), Turnover Costs: Evidence from Unexpected Worker Separations. Unpublished Manuscript.
- Bianchi, Nicola, Giulia Bovini, Jun Li, Matteo Paradisi and Michael Powell (2023), 'Careere Spillovers in Internal Labour Markets', *Review of Economic Studies* 90, 1800–1831.
- Bíró, Anikó, István Boza, Attila Gyetvai and Dániel Prinz (2024), Health Shocks, Social Insurance, and Firms, Discussion Paper No. 17069, Institute of Labor Economics, Bonn.
- Böheim, René (2017), The Labor Market in Austria, 2000-2016, IZA World of Labor 408, Institute for the Study of Labor, Bonn.
- Bouckaert, Nicolas, Anne C. Gielen and Tom Van Ourti (2020), 'It Runs in the Family Influenza Vaccination and Spillover Effects', *Journal of Health Economics* 74, 102386.
- Bradley, Steve, Colin Green and Gareth Leeves (2007), 'Worker Absence and Shirking: Evidence from Matched Teacher-School Data', *Labour Economics* 14, 319–334.

- Coile, Courtney C. (2004), Health Shocks and Couples' Labor Supply Decisions, Working Paper No. 10810, National Bureau of Economic Research, Cambridge, MA.
- Cornelissen, Thomas, Christian Dustmann and Uta Schönberg (2017), 'Peer Effects in the Workplace', American Economic Review 107(2), 425–456.
- Dachverband der Sozialversicherungsträger (2023), Statistisches Handbuch der österreichischen Sozialversicherung 2023, Dachverband der Sozialversicherungsträger, Wien.
- Dahl, Gordon B., Katrine V. Løken and Magne Mogstad (2014), 'Peer Effects in Program Participation', *American Economic Review* **104**(7), 2049–2074.
- Dobkin, Carlos, Amy Finkelstein, Raymond Kluender and Matthew J. Notowidigdo (2018), 'The Economic Consequences of Hospital Admissions', American Economic Review 108(2), 308–352.
- Eisenberg, Daniel, Ezra Golberstein and Janis L. Whitlock (2014), 'Peer Effects on Risky Behaviors: New Evidence From College Roommate Assignments', *Journal of Health Economics* **33**, 126–138.
- Fadlon, Itzik and Torben Heien Nielsen (2019), 'Family Health Behaviors', American Economic Review 109(9), 3162–3191.
- Fadlon, Itzik and Torben Heien Nielsen (2021), 'Family Labor Supply Responses to Severe Health Shocks: Evidence from Danish Administrative Records', American Economic Journal: Applied Economics 13(3), 1–30.
- Fietz, Katharina and Aiko Schmeißer (2024), Racial Peer Effects at Work. Evidence from Worker Deaths in Brazil, Policy Research Working Paper No. 10899, World Bank Group, Washington, D.C.
- Fletcher, Jason M. (2012), 'Peer Influences on Adolescent Alcohol Consumption: Evidence Using an Instrumental Variables/Fixed Effect Approach', Journal of Population Economics 25, 1265–1286.
- Francetic, Igor, Rachel Meacock and Matt Sutton (2022), 'Understanding Concordance in Health Behaviours Among Couples: Evidence From the Bowel Cancer Screening Programme in England', Journal of Economic Behavior and Organization 201, 310– 345.
- Frimmel, Wolfgang (2021), 'Later Retirement and the Labor Market Re-Integration of Elderly Unemployed Workers', *Journal of the Economics of Ageing* **19**, 100310.
- Frimmel, Wolfgang, Martin Halla, Jörg Paetzold and Julia Schmieder (forthcoming), 'Health of Parents, Their Children's Labor Supply, and the Role of Migrant Care Workers', *Journal of Labor Economics*.
- Frimmel, Wolfgang, Thomas Horvath, Mario Schnalzenberger and Rudolf Winter-Ebmer (2018), 'Seniority Wages and the Role of Firms in Retirement', *Journal of Public Economics* 164, 19–32.
- García-Gómez, Pilar, Hans van Kippersluis, Owen O'Donnell and Eddy van Doorslaer (2013), 'Long-Term and Spillover Effects of Health Shocks on Emplyoment and Income', *Journal of Human Resources* 48(5), 873–909.

- García-Gómez, Pillar (2011), 'Institutions, Health Shocks and Labour Market Outcomes Across Europe', *Journal of Health Economics* **30**, 200–213.
- Gaviria, Alejandro and Steven Raphael (2001), 'School-Based Peer Effects and Juvenile Behavior', *Review of Economics and Statistics* 83(2), 257–268.
- Glaser, Felix and Gerald J. Pruckner (2023), 'A Hard Pill to Swallow? Parental Health Shocks and Children's Mental Health', *Health Economics* **32**(12), 2768–2800.
- Goldberg, Jessica, Mario Macis and Pradeep Chintagunta (2023), 'Incentivized Peer Referrals for Tuberculosis Screening: Evidence From India', American Economic Journal: Applied Economics 15(1), 259–291.
- Goodman-Bacon, Andrew (2021), 'Difference-in-Differences With Variation in Treatment Timing', Journal of Econometrics **225**, 254–277.
- Guryan, Jonathan, Kory Kroft and Matthew J. Notowidigdo (2009), 'Peer Effects in the Workplace: Evidence from Random Groupings in Professional Golf Tournaments', *American Economic Journal: Applied Economics* 1(4), 34–68.
- Halla, Martin and Martina Zweimüller (2013), 'The Effect of Health on Earnings: Quasi-Experimental Evidence From Commuting Accidents', *Labour Economics* 24, 23–38.
- Hesselius, Patrik, Per Johansson and J. Peter Nilsson (2009), 'Sick of Your Colleagues' Absence?', Journal of the European Economic Association 7(2-3), 583–594.
- Ichino, Andrea and Giovanni Maggi (2000), 'Work Environment and Individual Background: Explaining Regional Shirking Differentials in a Large Italian Firm', *Quarterly Journal of Economics* 115(3), 1057–1090.
- Jäger, Simon and Jörg Heining (2022), How Substitutable are Workers? Evidence from Worker Deaths, Working Paper No. 30629, National Bureau of Economic Research, Cambridge, MA.
- Jaravel, Xavier, Neviana Petkova and Alex Bell (2018), 'Team-Specific Capital and Innovation', American Economic Review 108(4-5), 1034–1073.
- Jeon, Sung-Hee and R. Vincen Pohl (2017), 'Health and Work in the Family: Evidence from Spouses' Cancer Diagnoses', *Journal of Health Economics* 52, 1–18.
- Kapeller, Rudolf (2021), Shocks in the Workplace and Fimily Healthcare Utilization, Master Thesis, Department of Economics, Johannes Kepler University Linz, Linz.
- Maestas, Nicole, Matt Messel and Yulya Truskinovsky (2024), 'Caregiving and Labor Supply: New Evidence from Administrative Data', Journal of Labor Economics 42(S1), S183–S218.
- Markussen, Simen and Knut Røed (2015), 'Social Insurance Networks', Journal of Human Resources **50**(4), 1081–1113.
- Mas, Alexandre and Enrico Moretti (2009), 'Peers at Work', *American Economic Review* **99**(1), 112–145.

- Nahum, Ruth-Aïda (2007), Labour Supply Response to Spousal Sickness Absence, Arbetsrapport 2007:2, Institute for Future Studies, Stockholm.
- Norén, Anna (2020), Sick of My Parents? Consequences of Parental Ill Health on Adult Children, Working Paper No. 2020:1, Institute for Evaluation of Labour Market and Education Policy, Uppsala.
- OECD (2023), Pensions at a Glance 2023: OECD and G20 Indicators, OECD Publishing, Paris.
- Oettl, Alexander (2012), 'Reconceptualizing Stars: Scienti Helpfulness and Peer Performance', Management Science 58(6), 1122–1140.
- Parro, Francisco and R. Vincent Pohl (2021), 'The Effect of Accidents on Labor Market Outcomes: Evidence From Chile', *Health Economics* **30**, 1015–1032.
- Pollan, Wolfgang (2009), 'How Large are Wage Differentials in Austria?', Empirica 36, 389–406.
- Prinz, Daniel, Michael Chernew, David Cutler and Austin Frakt (2018), Health and Economic Activity Over the Lifecycle: Literature Review, Working Paper No. 24865, National Bureau of Economic Research, Cambridge, MA.
- Pruckner, Gerald J., Thomas Schober and Katrin Zocher (2020), 'The Company You Keep: Health Behavior Among Work Peers', *European Journal of Health Economics* 21, 251–259.
- Ramirez Lizardi, Eduardo, Elisabeth Fevang, Knut Røed and Henning Øien (2024), Effects of Health Shocks on Adult Children's Labor Market Outcomes and Well-Being, Discussion Paper No. 17232, Institute of Labor Economics, Bonn.
- Redler, Peter and Friederike Reichel (2024), When Do Peers Influence Preventitive Health Behavior? Evidence from Breast Cancer Screening. Unpublished Manuscript.
- Rellstab, Sara, Pieter Bakx, Pilar García-Gómez and Eddy van Doorslaer (2020), 'The Kids are Alright - Labour Market Effects of Unexpected Parental Hospitalisations in the Netherlands', *Journal of Health Economics* 69, 102275.
- Statistik Austria (2023), Zeitverwedung 2021/22. Ergebnisse der Zeitverwendungserhebung, Statistik Austria, Wien.
- Sun, Liyang and Sarah Abraham (2021), 'Estimating Dynamic Treatment Effects in Event Studies With Heterogeneous Treatment Effects', Journal of Econometrics 225, 175–199.
- Trevisan, Elisabetta and Francesca Zantomio (2016), 'The Impact of Acute Health Shocks on the Labour Supply of Older Workers: Evidence From Sixteen European Countries', *Labour Economics* 43, 171–185.
- Wu, Stephen (2003), 'The Effects of Health Events on the Economic Status of Married Couples', Journal of Human Resources XXXVIII(1), 219–230.
- Zweimüller, Josef, Rudolf Winter-Ebmer, Rafael Lalive, Andreas Kuhn, Jean-Philipe Wuellrich, Oliver Ruf and Simon Büchi (2009), Austrian Social Security Database, Working Paper No. 0903, The Austrian Center for Labor Economics and the Analysis of the Welfare State, Linz.

# 6 Figures (to be placed in the article)



Figure 1 Matching Diagnostics

*Note* — This figure shows several diagnostics for the propensity score matching described in Section 3.1. Panel (a) shows the share of matched worker-firm pairs over time. Panel (b) shows the distribution of propensity scores separately for treated shock workers (red solid line) and the matched control shock workers (blue dashed line). Panel (c) shows the distribution of the difference in propensity scores between shock workers in the treatment and matched control group.



Figure 2 Distribution of Health Shock Types

*Note* — The figure shows the distribution of the different types of severe health shocks in the main estimation sample. The health shocks are identified based on the admission diagnoses recorded in the hospital data. The first bar shows the share of health shocks associated with malignant neoplasms (ICD-10 C). The remaining bars show the share of different cardiovascular diseases, including cerebrovascular disease (ICD-10 I60 to I66), myocardial infarction (ICD-10 I21 and I22), and heart failure (ICD-10 I50).

(a) Inpatient Expenditures

(b) Inpatient Expenditures



Figure 3 Shock Workers – Healthcare Expenditures

*Note* — The first column of this figure shows sample averages with 95 percent confidence intervals for different types of shock workers' healthcare expenditures. The averages are shown separately for the treatment (red diamonds) and the control (blue dots) group. The second column shows the estimated coefficients with 95 percent confidence intervals for the effect of a severe health shock on different types of shock workers' healthcare expenditures. The estimates are based on Equation (2). Panels (a) and (b) consider the amount of inpatient expenditures in a given quarter. Panels (c) and (d) use outpatient expenditures as an outcome. Medication expenditures are considered in Panels (e) and (f). The estimates can be interpreted as the change in expenditures in a given quarter in Euros. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the health shock. The relative period -4 has been chosen as a reference period. In Panels (b), (d), and (f), the number of observations and the mean of the outcome variable for control shock workers in the reference period are reported in the bottom right corner. Standard errors are clustered at the shock firm level.

(a) Employment Days

(b) Employment Days



Figure 4 Shock Workers – Employment

Note — Panels (a) and (c) show sample averages with 95 percent confidence intervals for different employment outcomes of shock workers separately for the treatment (red diamonds) and the control (blue dots) group. The other panels show the estimated coefficients with 95 percent confidence intervals for the effect of a severe health shock. The estimates are based on Equation (2). Panels (a) and (b) consider the number of employment days in a given quarter. In Panels (c), (d), (e), and (f) we define a binary indicator that equals one if the shock workers is employed in the shock firm in a given quarter and zero otherwise. Panel (d) shows the coefficients for all shock workers in the estimation sample. In Panel (e) we estimate the regression separately for health shocks due to cancer (blue dots) and cardiovascular disease (gray diamonds). Panel (f) shows separate estimates for shock workers below (blue dots) and above (gray diamonds) the age of 50 at the time of the health shock. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the health shock. The relative period -4 has been chosen as a reference period. In Panels (b), (d), (e), and (f), the number of observations and the mean of the outcome variable for control shock workers in the reference period are reported in the bottom right corner. Standard errors are clustered at the shock firm level.

# 7 Tables (to be placed in the article)

	Ø Full	Ø Treatment	Ø Control	Diff.	Stand. Diff. in %	Ν
	(1)	(2)	(3)	(4)	(5)	(6)
Shock Firm Characteristics						
Firm Size	16.721	16.886	16.548	0.337	3.11	49,600
% Blue Collar	38.761	38.232	39.318	-1.086	-3.39	49,600
% Women	50.276	48.759	51.869	-3.110	-9.08	49,600
Avgerage Age	43.133	43.262	42.996	0.266	5.06	49,600
% Non-Austrian Citizens	7.513	7.110	7.936	-0.826	-6.52	49,600
Average Daily Wage	77.156	77.715	76.569	1.147	3.72	$49,\!600$
Coworker Characteristics						
Treated	0.512	1.000	0.000	$1.000^{***}$	0.00	$49,\!600$
Demographic Characteristics						
Female	0.516	0.504	0.528	-0.024	-4.77	49,600
Age in Years	52.992	52.943	53.045	-0.102	-2.85	49,600
Age at Shock in Years	54.572	54.523	54.624	-0.101	-2.91	49,600
Non-Austrian Citizen	0.063	0.061	0.065	-0.004	-1.71	$49,\!600$
Labor Market Outcomes						
Employment Days	88.736	88.765	88.705	0.060	0.44	49,600
Employed at Shock Firm	0.921	0.920	0.922	-0.002	-0.83	49,600
Blue Collar	0.433	0.424	0.443	-0.019	-3.82	48,807
Daily Wage	72.088	72.405	71.754	0.651	1.82	$48,\!173$
Tenure in Quarters	36.205	36.120	36.294	-0.173	-0.53	48,807
Health Outcomes						
Sick Leave Days (All Causes)	2.357	2.277	2.442	-0.164	-2.00	48,950
Easy-to-Fake Sick Leave Days	0.114	0.109	0.118	-0.009	-0.62	48,934
Inpatient Exp.	95.907	93.676	98.255	-4.579	-0.59	48,321
Outpatient Exp.	94.230	92.933	95.596	-2.663	-1.57	48,321
Medication Exp.	30.099	25.547	34.893	$-9.346^{***}$	-6.94	48,321
Exp. Preventative Health Check-Up	2.910	2.858	2.965	-0.107	-0.73	48,935
Exp. PSA Screening	0.486	0.476	0.498	-0.022	-1.10	23,561
Exp. Mammography	4.079	4.021	4.136	-0.116	-0.72	25,381
Dermatology Exp.	1.668	1.585	1.756	-0.171	-1.57	48,321
Radiology Exp.	5.183	5.025	5.350	-0.326	-1.61	48,321

#### Table 1 Comparing Treatment and Control Coworkers

Note — The table reports descriptive statistics for shock firm characteristics as well as the demographic, labor market, and health characteristics of the coworkers included in our main estimation sample. The definition of the main estimation sample and the construction of the treatment and control group are described in Section 3.1. The sample period comprises the 12 quarters preceding the (placebo) health shock. Note that health characteristics can only be observed for coworkers insured in Upper Austria. Prostate-specific antigen (PSA) tests are analyzed only for men, while mammographies are analyzed only for women. Column (1) reports the mean for the full sample. The means for treated and control coworkers are reported in Columns (2) and (3). Column (4) reports the difference in means between the treatment and control group. Column (5) reports the standardized difference in means. The standardized difference is defined as the difference in means between the treated and control group for a given variable ( $\mu_{treated}-\mu_{control}$ ) divided by the average standard deviation ( $\sqrt{0.5 \cdot (\sigma_{treated}^2 + \sigma_{control}^2)}$ ) and multiplied by 100. The number of observations is given in Column (6). Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

	En	nployment	Retirement	Wag	Wages		
	Days	P(Employed at Shock Firm)	Days	Log Daily Wage	Wage Growth		
	(1)	(2)	(3)	(4)	(5)		
Health Shock $\times$ Post	$1.0676^{*}$ (0.6159)	$0.0232^{*}$ (0.0139)	$-1.0506^{**}$ (0.4612)	$0.0022 \\ (0.0059)$	-0.0002 (0.0015)		
Individual FE Year × Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	$93,058 \\ 1,605 \\ 0.218 \\ 88.705 \\ 1.204$	$\begin{array}{c} 93,058 \\ 1,605 \\ 0.541 \\ 0.922 \\ 2.520 \end{array}$	93,058 1,605 0.246 7.829 -13.419	84,129 1,604 0.687 4.130 0.053	79,482 1,604 0.579 0.006 -3.746		

 Table 2
 Average Effects of Health Shocks on Coworkers' Labor Market Outcomes

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' labor market outcomes. The results are based on Equation (1). Column (1) shows the results for the number of employment days in a given quarter. Column (2) uses a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Column (3) reports the results for the quarterly number of retirement days. Column (4) uses the log daily wage as an outcome. Column (5) considers wage growth, which is defined as the difference in log daily wages between two consecutive quarters, as an outcome. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

	Sick Le	eave Days	$\mathbf{P}(\text{Expenditures} > 0)$				
	All Causes	$\begin{array}{c} {\rm Causes} & {\rm Easy-to-Fake} & {\rm General\ Hea} \\ {\rm Diagnoses} & {\rm Check-UI} \end{array}$		PSA Screening	Mammography		
	(1)	(2)	(3)	(4)	(5)		
Health Shock $\times$ Post	-0.1002 (0.1912)	0.0043 (0.0283)	0.0009 (0.0022)	0.0044 (0.0044)	$0.0064^{*}$ (0.0034)		
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	91,862 1,605 0.009 2.442 -4.103	$91,830 \\ 1,605 \\ 0.442 \\ 0.118 \\ 3.593$	$91,829 \\ 1,605 \\ 0.293 \\ 0.040 \\ 2.186$	$\begin{array}{c} 43,788\\ 958\\ 0.061\\ 0.060\\ 7.300\end{array}$	$\begin{array}{c} 48,050\\ 1,128\\ 0.645\\ 0.065\\ 9.726\end{array}$		

Table 3 Average Effects of Health Shocks on Coworkers' Health Outcomes

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' health outcomes. The results are based on Equation (1). Column (1) shows the results for the number of sick leave days in a given quarter. Column (2) considers only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). In Column (3), we define a binary indicator that equals one if the outpatient expenditures for the general health screening program in a given quarter are larger than zero and zero otherwise. Columns (4) and (5) use analogous binary indicators for expenditures related to PSA tests and mammography. These outcomes are analyzed only for men and women, respectively. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome wariable for the control group in the pre-treatment period, and the coefficient in percent of the outcome mean are reported at the bottom of the table. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Emplo	oyment		Reti	rement	$\mathbf{P}(\mathrm{Expend})$	litures $> 0$ )
	D	ays	P(Em at Shoo	ployed k Firm)	I	Days	Mamm	ography
Wage Percentile of Shock Worker	$\leq 75^{\mathrm{th}}$	$> 75^{\mathrm{th}}$	$\leq 75^{\mathrm{th}}$	$> 75^{\mathrm{th}}$	$\leq 75^{\mathrm{th}}$	$> 75^{\mathrm{th}}$	$\leq 75^{\mathrm{th}}$	$> 75^{\mathrm{th}}$
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Health Shock $\times$ Post	$\begin{array}{c} 0.4024 \\ (0.7581) \end{array}$	$2.5464^{**} \\ (1.0516)$	$0.0080 \\ (0.0161)$	$0.0586^{**}$ (0.0267)	-0.4244 (0.5487)	$-2.2705^{***}$ (0.7921)	$0.0076^{*}$ (0.0042)	0.0048 (0.0059)
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	$\begin{array}{c} 61,342 \\ 1,039 \\ 0.232 \\ 88.419 \\ 0.455 \end{array}$	31,715 570 0.471 89.179 2.855	$\begin{array}{c} 61,342 \\ 1,039 \\ 0.203 \\ 0.921 \\ 0.869 \end{array}$	31,715 570 0.498 0.925 6.337	61,342 1,039 0.102 7.310 -5.805	31,715 570 0.488 8.590 -26.432	30,729 734 0.181 0.066 11.515	$17,321 \\ 397 \\ 0.945 \\ 0.064 \\ 7.526$

Table 4 Average Effects of Health Shocks on Coworkers' Outcomes by Wage Percentile of Shock Worker

*Note* — The table reports the estimated coefficients for the effect of a health shock on coworkers' outcomes by the within-firm wage percentile of the shock worker at the time of the health shock. We estimate Equation (1) separately for coworkers of shock workers with daily earnings below the 75<sup>th</sup> percentile of the wage distribution of the shock firm and for coworkers of shock workers with daily earnings above the 75<sup>th</sup> percentile. Columns (1), (3), (5), and (7) report the results for the first group, Columns (2), (4), (6), and (8) those for the second group. Columns (1) and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. In Columns (7) and (8), we define a binary indicator that equals one if the outpatient expenditures for mammography in a given quarter are larger than zero and zero otherwise. This outcome is only analyzed for women. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Employ	yment		Ret	rement	$\mathbf{P}(\mathrm{Expend})$	$\mathbf{P}(\text{Expenditures} > 0)$		
	Γ	Days	P(Em at Shoc	ployed k Firm)	I	Days		Mammography		
Productivity of Shock Worker	Low (1)	High (2)	Low (3)	High (4)	Low (5)	High (6)	Low (7)	High (8)		
Health Shock $\times$ Post	0.2493 (0.8641)	$2.4080^{***}$ (0.8401)	$\begin{array}{c} 0.0132\\ (0.0178) \end{array}$	$0.0371^{*}$ (0.0213)	-0.3556 (0.6328)	$-2.1753^{***}$ (0.6143)	0.0062 (0.0044)	$0.0058 \\ (0.0054)$		
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	$\begin{array}{r} 49,322\\ 865\\ 0.033\\ 88.136\\ 0.283\end{array}$	$\begin{array}{r} 43,472 \\ 736 \\ 0.403 \\ 89.292 \\ 2.697 \end{array}$	$\begin{array}{r} 49,322\\ 865\\ 0.564\\ 0.921\\ 1.430\end{array}$	$\begin{array}{r} 43,472 \\ 736 \\ 0.843 \\ 0.923 \\ 4.022 \end{array}$	$\begin{array}{r} 49,322\\ 865\\ 0.213\\ 7.891\\ -4.506\end{array}$	43,472 736 0.949 7.726 -28.155	29,195 650 0.646 0.062 9.955	$18,714 \\ 474 \\ 0.837 \\ 0.069 \\ 8.340$		

**Table 5** Average Effects of Health Shocks on Coworkers' Outcomes by Productivity of Shock Worker

*Note* — The table reports the estimated coefficients for the effect of a health shock on coworkers' outcomes by the productivity of the shock worker. We estimate Equation (1) separately for coworkers of shock workers with a below median AKM worker fixed effect and for coworkers of shock workers with an above median AKM worker fixed effect. Columns (1), (3), (5), and (7) report the results for the first group, Columns (2), (4), (6), and (8) those for the second group. Columns (1) and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. In Columns (7) and (8), we define a binary indicator that equals one if the outpatient expenditures for mammography in a given quarter are larger than zero and zero otherwise. This outcome is analyzed only for women. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

# Web Appendix

This Web Appendix provides additional material discussed in the unpublished manuscript "Sick Happens: The Effect of Worker Health Shocks on Coworkers' Employment and Health Behavior" by Wolfgang Frimmel and Rene Wiesinger.

# A Additional Figures and Tables

## A.1 Additional Figures



(b) Expenditures for Mammography



Figure A.1 Shock Workers – Preventive Healthcare

*Note* — The figure shows the estimated coefficients with 95 percent confidence intervals for different types of shock workers' preventive healthcare expenditures. The estimates are based on Equation (2). Panel (a) considers the amount of outpatient expenditures for prostate-specific antigen (PSA) tests in a given quarter. Panel (b) uses expenditures for mammography as an outcome. The outcomes in Panels (a) and (b) are analyzed only for men and women, respectively. Expenditures for dermatologists are considered in Panel (c). Panel (d) shows the change in the amount of expenditures for radiologists. The estimates can be interpreted as the change in expenditures in a given quarter in Euros. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the health shock. The relative period -4 has been chosen as a reference period. The number of observations and the mean of the outcome variable for control shock workers in the reference period are reported in the bottom right corner. Standard errors are clustered at the shock firm level.



(a) Overall

Figure A.2 Shock Workers – Mortality

*Note* — The figure shows sample averages with 95 percent confidence intervals for shock workers' mortality. Mortality is defined as the probability of dying within a given number of quarters, i.e., the cumulative probability of dying before a given quarter (including the respective quarter). The averages are shown separately for the treatment (red diamonds) and the control (blue dots) group. Panel (a) shows the overall probabilities for the treatment and control group. In Panel (b), probabilities for the treatment group are shown separately for severe health shocks due to cancer (red squares) and health shocks due to acute cardiovascular events (red diamonds). Panel (c) differentiates between treated shock workers below the age of 50 at the time of the health shock (red squares) and those above the age of 50 (red diamonds). Averages are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the health shock.



Figure A.3 Shock Workers – Sick Leave Days and Retirement

Note — The figure shows the estimated coefficients with 95 percent confidence intervals for shock workers' sick leave days and retirement days. The estimates are based on Equation (2). Panel (a) shows the results for the number of sick leave days in a given quarter. Panel (b) includes only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). Panel (c) shows the change in the quarterly number of retirement days. Panel (d) considers only days associated with disability retirement. The estimates can be interpreted as the change in the number of days in a given quarter. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the health shock. The relative period -4 has been chosen as a reference period. The number of observations and the mean of the outcome variable for control shock workers in the reference period are reported in the bottom right corner. Standard errors are clustered at the shock firm level.







*Note* — The figure shows the estimated coefficients with 95 percent confidence intervals for coworkers' labor market outcomes. The estimates are based on Equation (2). Panel (a) shows the results for the number of employment days in a given quarter. Panel (b) uses a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Panel (c) shows the change in the quarterly number of retirement days. Panel (d) uses the log daily wage as an outcome. Panel (e) considers wage growth, which is defined ans the difference in log daily wages between two consecutive quarters, as an outcome. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the (placebo) health shock. The relative period -4 has been chosen as a reference period. The number of observations and the mean of the outcome wariable for on observations from the relative period -4. The outcome mean for retirement is based on the relative period 4. Standard errors are clustered at the shock firm level.

(a) All Sick Leave Days

(b) Easy-to-Fake Sick Leave Days



#### (c) $\mathbf{P}(\text{Exp. General Health Check-Up} > 0)$



(d)  $\mathbf{P}(\text{Exp. PSA Screening} > 0)$ 



(e)  $\mathbf{P}(\text{Exp. Mammography} > 0)$ 





Note — The figure shows the estimated coefficients with 95 percent confidence intervals for coworkers' health outcomes. The estimates are based on Equation (2). Panel (a) shows the results for the number of sick leave days in a given quarter. Panel (b) considers only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). In Panel (c), we define a binary indicator that equals one if the outpatient expenditures for the general health screening program in a given quarter are larger than zero and zero otherwise. Panels (d) and (e) use analogous binary indicators for expenditures related to PSA tests and mammography. These outcomes are analyzed only for men and women, respectively. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the (placebo) health shock. The relative period -4 has been chosen as a reference period. The number of observations and the mean of the outcome variable for the control group in the reference period are reported in the bottom right corner. Standard errors are clustered at the shock firm level.

(a) Employment Days

(b) **P**(Employed at Shock Firm)



# **Figure A.6** Dynamic Effects of Health Shocks on Coworkers' Labor Market Outcomes – Alternative Estimation Method

Note — The figure shows the estimated coefficients with 95 percent confidence intervals for coworkers' labor market outcomes for different estimation methods. The estimates are based on Equation (2). Estimates represented by blue dots are based on the estimation method by Sun and Abraham (2021). Gray diamonds represent the baseline two-way fixed effects estimates. Panel (a) shows the results for the number of employment days in a given quarter. Panel (b) uses a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Panel (c) shows the change in the quarterly number of retirement days. Panel (d) uses the log daily wage as an outcome. Panel (e) considers wage growth, which is defined ans the difference in log daily wages between two consecutive quarters, as an outcome. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the (placebo) health shock. The relative period -4 has been chosen as a reference period. The number of observations and the mean of the outcome variable for the control group are reported in the bottom right corner. For all outcomes except retirement, the outcome mean is based on observations from the relative period -4. The outcome mean for retirement is based on the relative period 4. Standard errors are clustered at the shock firm level. (a) All Sick Leave Days





(c)  $\mathbf{P}(\text{Exp. General Health Check-Up} > 0)$ 





(d)  $\mathbf{P}(\text{Exp. PSA Screening} > 0)$ 



(e)  $\mathbf{P}(\text{Exp. Mammography} > 0)$ 



# **Figure A.7** Dynamic Effects of Health Shocks on Coworkers' Health Outcomes – Alternative Estimation Method

*Note* — The figure shows the estimated coefficients with 95 percent confidence intervals for coworkers' health outcomes for different estimation methods. The estimates are based on Equation (2). Estimates represented by blue dots are based on the estimation method by Sun and Abraham (2021). Gray diamonds represent the baseline two-way fixed effects estimates. The estimates are based on Equation (2). Panel (a) shows the results for the number of sick leave days in a given quarter. Panel (b) considers only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). In Panel (c), we define a binary indicator that equals one if the outpatient expenditures for the general health screening program in a given quarter are larger than zero and zero otherwise. Panels (d) and (e) use analogous binary indicators for expenditures related to PSA tests and mammography. These outcomes are analyzed only for men and women, respectively. Estimates are shown for 12 quarters before and 12 quarters (including the quarter of the health shock) after the health shock. Zero indicates the quarter of the (placebo) health shock. The relative period -4 has been chosen as a reference period. The number of observations and the mean of the outcome variable for the control group in the reference period are reported in the bottom right corner. Standard errors are clustered at the shock firm level.

## A.2 Additional Tables

	Ø All Obs.	Ø Matched	Ø Non-Matched	Diff.	Stand. Diff. in %	Ν
	(1)	(2)	(3)	(4)	(5)	(6)
Firm Characteristics						
Firm Size	16.996	11.897	21.153	$-9.256^{***}$	-92.45	2,302
% Women	46.470	52.265	41.744	$10.521^{***}$	32.53	2,302
% Blue Collar	40.473	36.112	44.028	$-7.916^{***}$	-24.74	2,302
% Non-Austrian Citizens	8.460	6.243	10.269	$-4.026^{***}$	-30.39	2,302
Average Worker Age	40.681	40.793	40.590	0.203	3.61	2,302
Avgerage Daily Wage	79.277	75.148	82.645	$-7.496^{***}$	-28.21	2,302
Firm Age	17.534	18.343	16.874	$1.470^{***}$	11.08	2,302
Worker Characteristics						
Matched	0.449	1.000	0.000	$1.000^{***}$	0.00	2,302
$Demographic \ Characteristics$						
Age	48.017	47.648	48.317	$-0.669^{*}$	-7.36	2,302
Female	0.438	0.512	0.378	$0.134^{***}$	27.17	2,302
Non-Austrian Citizen	0.079	0.025	0.124	$-0.099^{***}$	-38.25	2,302
Labor Market Outcomes						
Employed at Shock Firm	0.878	0.952	0.818	$0.134^{***}$	42.84	2,302
Employment Days	89.819	90.044	89.608	0.435	5.14	2,137
Blue Collar	0.453	0.416	0.488	$-0.072^{***}$	-14.48	2,137
Daily Wage	73.556	69.136	77.703	$-8.566^{***}$	-24.95	2,136
Unemployment Days	1.405	1.145	1.648	-0.503	-5.67	2,137
Retirement Days	0.556	0.355	0.745	-0.390	-5.52	2,137
Health Outcomes						
Sick Leave Days (All Causes)	2.053	2.049	2.056	-0.007	-0.11	2,114
Hospital Days	0.156	0.173	0.141	0.032	3.13	1,977
Outpatient Expenditures	90.827	87.797	93.665	-5.868	-3.22	1,977
Medication Expenditures	44.490	38.514	50.086	-11572	-4.80	1,977
GP Expenditures	23.147	23.271	23.030	0.241	0.68	1,977

 Table A.1
 Comparing Matched and Non-Matched Worker-Firm Pairs

Note — The table reports descriptive statistics for firm characteristics as well as the demographic, labor market, and health characteristics separately for worker-firm pairs with a health shock that could and could not be matched with worker-firm pairs from the pool of worker-firm pairs without a health shock. The sample includes all worker-firms pairs for which a health shock was identified as described in Section 3.1. We include only firms that have fewer than 50 workers four quarters before the health shock. Note that health characteristics can only be observed for coworkers insured in Upper Austria. Column (1) reports the mean for the full sample. The means for matched and non-matched worker-firm pairs are reported in Columns (2) and (3). Column (4) reports the difference in means between the treatment and control group. Column (5) reports the standardized difference in means. The standardized difference is defined as the difference in means between the treated and control group for a given variable ( $\mu_{treated} - \mu_{control}$ ) divided by the average standard deviation ( $\sqrt{0.5 \cdot (\sigma_{treated}^2 + \sigma_{control}^2)}$ ) and multiplied by 100. The number of observations is given in Column (6). Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

	Ø Full	Ø Treatment	Ø Control	Diff.	Stand. Diff. in $\%$	Ν
	(1)	(2)	(3)	(4)	(5)	(6)
Shock Worker Characteristics						
Treated	0.496	1.000	0.000	$1.000^{***}$	0.00	18,221
Demographic Characteristics						
Female	0.510	0.501	0.519	-0.018	-3.67	18,221
Age in Years	47.625	47.853	47.401	0.452	5.56	18,221
Age at Shock in Years	49.201	49.435	48.971	0.464	5.72	18,221
Non-Austrian Citizen	0.026	0.025	0.027	-0.002	-1.42	18,221
Labor Market Outcomes						
Employment Days	88.573	88.643	88.505	0.138	0.98	18,221
Employed at Shock Firm	0.915	0.919	0.912	0.007	2.68	18,221
Blue Collar	0.422	0.414	0.430	-0.015	-3.05	17,906
Daily Wage	69.724	69.944	69.507	0.438	1.30	17,707
Tenure in Quarters	32.442	31.850	33.026	-1.176	-3.90	$17,\!906$
Health Outcomes						
Sick Leave Days (All Causes)	2.260	2.326	2.195	0.131	1.68	18,067
Easy-to-Fake Sick Leave Days	0.100	0.081	0.118	-0.037	-2.73	18,046
Inpatient Exp.	110.687	125.878	96.666	$29.211^{*}$	2.96	17,259
Outpatient Exp.	87.760	91.070	84.704	6.366	3.87	17,259
Medication Exp.	35.831	43.893	28.389	15.504	7.27	17,259
Exp. Preventative Health Check-Up	2.431	2.456	2.406	0.050	0.37	18,046
Exp. PSA Screening	0.383	0.436	0.328	$0.108^*$	5.78	8,810
Exp. Mammography	4.008	4.025	3.992	0.032	0.20	9,237
Dermatology Exp.	2.010	2.238	1.800	0.438	3.52	17,259
Radiology Exp.	4.916	5.039	4.802	0.237	1.19	$17,\!259$

 Table A.2
 Comparing Treatment and Control Shock Workers

Note — The table reports descriptive statistics for the demographic, labor market, and health characteristics of the shock workers included in our main estimation sample. The definition of the main estimation sample and the construction of the treatment and control group are described in Section 3.1. The sample period comprises the 12 quarters preceding the (placebo) health shock. Note that health characteristics can only be observed for shock workers insured in Upper Austria. Prostate-specific antigen (PSA) tests are analyzed only for men, while mammographies are analyzed only for women. Column (1) reports the mean for the full sample. The means for treated and control shock workers are reported in Columns (2) and (3). Column (4) reports the difference in means between the treatment and control group. Column (5) reports the standardized difference in means. The standardized difference is defined as the difference in means between the treated and control group for a given variable ( $\mu_{treated} - \mu_{control}$ ) divided by the average standard deviation ( $\sqrt{0.5 \cdot (\sigma_{treated}^2 + \sigma_{control}^2)}$ ) and multiplied by 100. The number of observations is given in Column (6). Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

	Ν	% Overall	% Within Category
	(1)	(2)	(3)
Malignant Neoplasms			
m C01-C14	15	1.88	2.74
C15 - C25	59	7.39	10.77
C31 - C34	33	4.14	6.02
C40 - C41	3	0.38	0.55
C43 - C44	90	11.28	16.42
C48 - C49	8	1.00	1.46
C50	124	15.54	22.63
C51 - C57	44	5.51	8.03
m C60- m C62	62	7.77	11.31
C64 - C67	24	3.01	4.38
C69 - C71	14	1.75	2.55
C73 - C75	20	2.51	3.65
C76 - C80	18	2.26	3.28
C81 - C96	34	4.26	6.20
Cerebrovascular Diseases			
I60 - I65	147	18.42	100.00
Myocardial Infarction			
I21	80	10.03	98.77
I22	1	0.13	1.23
Heart Failure			
I50	22	2.76	100.00

 Table A.3 Distribution of Shock Diagnoses

Note — The table reports the distribution of the different hospital admission diagnosis groups used to identify severe health shocks. Column (1) reports the absolute number of health shocks for each diagnosis group. Column (2) shows the percentage share of each diagnosis group in the total number of health shocks. Column (3) reports the percentage share of each diagnosis group within the four different types of health shocks (malignant neoplasms, cerebrovascular diseases, myocardial infarction, and heart failure). The ICD-10 codes represent the following diagnoses: C01 - C14 – malignant neoplasms of lip, oral cavity, and pharynx, C15 - C25 - malignant neoplasms of digestive organ, C31 - C34 malignant neoplasms of respiratory and intrathoracic organs, C40 - C41 - malignant neoplasms of bone and articular cartilage, C43 – C44 – melanoma and other malignant neoplasms of skin, C48 – C49 – malignant neoplasms of mesothelial and soft tissue, C50 – malignant neoplasm of breast, C51 – C57 – malignant neoplasms of female genital organs, C60 - C62 - malignant neoplasms of male genital organs, C64 - C67 - malignant neoplasms of urinary tract, C69 - C71 - malignant neoplasms of eye, brain and other parts of central nervous system, C73 - C75 - malignant neoplasms of thyroid and other endocrine glands, C76 – C80 – malignant neoplasms of ill-defined, secondary and unspecified sites, C81 – C96 – malignant neoplasms, stated or presumed to be primary, of lymphoid, haematopoietic and related tissue, I60 – I65 – cerebrovascular diseases, I21 – acute myocardial infarction, I22 – subsequent myocardial infarction, I50 – heart failure.

	All Obs	ervations	Coworker the Labo	s Leaving r Market
	(1)	(2)	(3)	(4)
Health Shock	$\begin{array}{c} 0.0101^{**} \\ (0.0047) \end{array}$	$0.0105^{**}$ (0.0046)	$0.0349^{**}$ (0.0162)	$0.0300^{*}$ (0.0173)
Coworker Characteristics				
Non-Austrian Citizen		0.0096 (0.0140)		0.0269 (0.0354)
Female		-0.0002 (0.0050)		-0.0146 (0.0225)
Blue Collar		-0.0005 (0.0059)		-0.0020 (0.0196)
Shock Worker Characteristic	s			
Non-Austrian Citizen		$-0.0186^{*}$		-0.0378
Female		(0.0100) -0.0060 (0.0064)		(0.0292) -0.0319 (0.0285)
Blue Collar		(0.0064) -0.0065 (0.0070)		(0.0285) -0.0322 (0.0266)
		(0.0070)		(0.0200)
Shock Year FE		$\checkmark$		$\checkmark$
Coworker Age at Shock FE Shock Worker Age at Shock FE		$\checkmark$		$\checkmark$
N	1,564	1,564	435	435
# Clusters	632	632	299	299
Outcome Mean	0.0026	0.0026	0.0095	0.0095

#### Table A.4 Effect of Health Shocks on Coworkers' Probability of Receiving a Golden Handshake

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' probability of receiving a golden handshake after the health shock. The results are based on a cross-section of all coworkers included in the main estimation sample. The outcome variable is a binary indicator that is one if a coworker receives a special severance payment that is at least 8 times as large as their monthly earnings at some point after the health shock and zero otherwise. Columns (1) and (2) include all coworkers in the main estimation sample, while Columns (3) and (4) include only those coworkers whose last employment spell before retirement is at the shock firm. As the information on severance pay is only available until 2012, the sample only includes health shocks before 2010. The regressions for Columns (1) and (3) only include a binary treatment indicator as an explanatory variable. In Columns (2) and (4) we additionally control for coworkers' and shock workers' age, citizenship, gender, and collar. We also add a fixed effect for the year of the shock. The number of observations, the number of clusters, and the mean of the outcome variable for the control group are reported at the bottom of the table. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

	En	nployment	Retirement	Wag	Wages		
	Days	P(Employed at Shock Firm)	Days	Log Daily Wage	Wage Growth		
	(1)	(2)	(3)	(4)	(5)		
Health Shock $\times$ Post	$0.7235^{*}$ (0.4122)	$0.0101 \\ (0.0128)$	$\begin{array}{c} 0.1233^{***} \\ (0.0401) \end{array}$	$0.0052 \\ (0.0040)$	0.0004 (0.0009)		
Individual FE Year × Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	$263,205 \\ 1,949 \\ 0.400 \\ 83.014 \\ 0.872$	$263,205 \\ 1,949 \\ 0.656 \\ 0.829 \\ 1.216$	$263,205 \\ 1,949 \\ 0.630 \\ 0.027 \\ 459.269$	$\begin{array}{c} 238,613 \\ 1,949 \\ 0.146 \\ 4.092 \\ 0.128 \end{array}$	$\begin{array}{c} 224,619\\ 1,949\\ 0.202\\ 0.013\\ 2.817\end{array}$		

 Table A.5
 Average Effects of Health Shocks on Young Coworkers' Labor Market Outcomes

Note — The table reports the estimated coefficients for the effect of a health shock on young coworkers' labor market outcomes. The estimation sample includes only coworkers who are below the age of 50 at the time of the (placebo) health shock. The results are based on Equation (1). Column (1) shows the results for the number of employment days in a given quarter. Column (2) uses a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Column (3) reports the results for the quarterly number of retirement days. Column (4) uses the log daily wage as an outcome. Column (5) considers wage growth, which is defined as the difference in log daily wages between two consecutive quarters, as an outcome. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

	Sick Le	eave Days	Р	P(Expenditures > 0)	0)	
	All Causes	Easy-to-Fake Diagnoses	General Health Check-Up	PSA Screening	Mammography	
	(1)	(2)	(3)	(4)	(5)	
Health Shock $\times$ Post	-0.0544 (0.0866)	0.0054 (0.0097)	0.0017 (0.0011)	-0.0012 (0.0010)	-0.0002 (0.0016)	
Individual FE Year × Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	255,274 1,949 0.252 1.921 -2.830	$254,931 \\ 1,949 \\ 0.171 \\ 0.073 \\ 7.374$	$\begin{array}{c} 254,951 \\ 1,949 \\ 0.764 \\ 0.025 \\ 6.916 \end{array}$	$\begin{array}{c} 139,702 \\ 1,398 \\ 0.363 \\ 0.007 \\ -16.909 \end{array}$	$115,230 \\ 1,639 \\ 0.997 \\ 0.027 \\ -0.581$	

 Table A.6
 Average Effects of Health Shocks on Young Coworkers' Health Outcomes

Note — The table reports the estimated coefficients for the effect of a health shock on young coworkers' health outcomes. The estimation sample includes only coworkers who are below the age of 50 at the time of the (placebo) health shock. The results are based on Equation (1). Column (1) shows the results for the number of sick leave days in a given quarter. Column (2) includes only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). In Column (3), we define a binary indicator that equals one if the outpatient expenditures for the general health screening program in a given quarter are larger than zero and zero otherwise. Columns (4) and (5) use analogous binary indicators for expenditures related to PSA tests and mammography. These outcomes are analyzed only for men and women, respectively. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome wariable for the control group in the pre-treatment period, and the coefficient in percent of the outcome mean are reported at the bottom of the table. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Emplo	yment		Retire	ment	$\mathbf{P}(\text{Expenditures} > 0)$	
	Da	iys	P(Em at Shoc	ployed k Firm)	Days		Mammography	
Age of Shock Worker	< 50	$\geq 50$	< 50	$\geq 50$	< 50	$\geq 50$	< 50	$\geq 50$
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Health Shock $\times$ Post	$1.9076^{**}$ (0.9002)	$0.7358 \\ (0.8489)$	$0.0332^{*}$ (0.0191)	$\begin{array}{c} 0.0202\\ (0.0196) \end{array}$	$-1.7430^{***}$ (0.6467)	-0.6019 (0.6393)	$\begin{array}{c} 0.0045 \\ (0.0054) \end{array}$	$0.0095^{**}$ (0.0044)
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
N	40,743	52,315	40,743	52,315	40,743	52,315	20,799	27,251
# Clusters	722	884	722	884	722	884	499	630
p-Value of Pre-Coefficients	0.754	0.083	0.921	0.474	0.507	0.827	0.751	0.848
Outcome Mean	88.505	88.882	0.917	0.927	8.839	6.883	0.070	0.062
Effect in % of Mean	2.155	0.828	3.626	2.181	-19.719	-8.745	6.512	15.400

Table A.7 Average Effects of Health Shocks on Coworkers' Outcomes by Age of Shock Worker

*Note* — The table reports the estimated coefficients for the effect of a health shock on coworkers' outcomes by the age of the shock worker at the time of the (placebo) health shock. We estimate Equation (1) separately for coworkers of shock workers below the age of 50 and for coworkers of shock workers above the age of 50. Columns (1), (3), (5), and (7) report the results for the first group, Columns (2), (4), (6), and (8) those for the second group. Columns (1) and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. In Columns (7) and (8), we define a binary indicator that equals one if the outpatient expenditures for mammography in a given quarter are larger than zero and zero otherwise. This outcome is analyzed only for women. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Emplo	oyment		Reti	rement	$\mathbf{P}(\text{Expenditures} > 0)$		
	Days		<b>P</b> (Employed at Shock Firm)		Ē	ays	Mammography		
Age of Coworker	< 55(1)	$ \geq 55 \\ (2) $	< 55(3)	$ \geq 55 \\ (4) $	< 55(5)	$ \geq 55 $ (6)	< 55 (7)	$\geq 55$ (8)	
Health Shock $\times$ Post	$0.5673 \\ (0.5708)$	$\begin{array}{c} 1.6420 \\ (1.1715) \end{array}$	$\begin{array}{c} 0.0113 \\ (0.0155) \end{array}$	$0.0341^{*}$ (0.0192)	-0.2006 (0.1623)	$-2.1280^{**}$ (1.0156)	$0.0076^{*}$ (0.0046)	0.0049 (0.0051)	
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	51,097 1,243 0.699 88.696 0.640	41,961 1,123 0.275 88.717 1.851	$51,097 \\ 1,243 \\ 0.512 \\ 0.920 \\ 1.232$	$\begin{array}{c} 41,961 \\ 1,123 \\ 0.703 \\ 0.925 \\ 3.682 \end{array}$	51,097 1,243 0.709 0.434 -46.198	$\begin{array}{c} 41,961\\ 1,123\\ 0.158\\ 16.963\\ -12.545\end{array}$	$27,041 \\ 802 \\ 0.131 \\ 0.069 \\ 11.010$	$21,009 \\ 668 \\ 0.656 \\ 0.061 \\ 8.159$	

Table A.8 Average Effects of Health Shocks on Coworkers' Outcomes by Age of Coworker

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' outcomes by the age of the coworker at the time of the (placebo) health shock. We estimate Equation (1) separately for coworkers aged between 50 and 54 and for coworkers above the age of 55. Columns (1), (3), (5), and (7)report the results for the first group, Columns (2), (4), (6), and (8) those for the second group. Columns (1)and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. In Columns (7) and (8), we define a binary indicator that equals one if the outpatient expenditures for mammography in a given quarter are larger than zero and zero otherwise. This outcome is analyzed only for women. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Emplo	Retirement				
	Da	ays	P(Em at Shoc	ployed k Firm)	Days		
Gender of Coworker	Women (1)	Men (2)	Women (3)	Men (4)	Women (5)	Men (6)	
Health Shock $\times$ Post	$0.4150 \\ (0.8440)$	$1.6689^{**}$ (0.8251)	0.0074 (0.0160)	$0.0434^{**}$ (0.0205)	$-1.4157^{**}$ (0.5970)	-0.2086 (0.5059)	
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	$\begin{array}{c} 48,436\\ 1,128\\ 0.043\\ 88.669\\ 0.468\end{array}$	$\begin{array}{r} 44,\!622\\958\\0.135\\88.746\\1.881\end{array}$	$\begin{array}{c} 48,436\\ 1,128\\ 0.647\\ 0.925\\ 0.796\end{array}$	$\begin{array}{c} 44,622\\958\\0.326\\0.919\\4.723\end{array}$	$\begin{array}{c} 48,\!436\\ 1,\!128\\ 0.233\\ 9.766\\ -14.496\end{array}$	$\begin{array}{r} 44,622\\ 958\\ 0.106\\ 5.600\\ -3.725\end{array}$	

#### Table A.9 Average Effects of Health Shocks on Coworkers' Outcomes by Gender of Coworker

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' outcomes by the the gender of the coworker. We estimate Equation (1) separately for women and men. Columns (1), (3), and (5) report the results for the first group, Columns (2), (4), and (6) those for the second group. Columns (1) and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. Since the probability of having a mammography is analyzed only for women, this outcomes is omitted from this table. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Emplo	yment		Retir	ement	$\mathbf{P}(\mathbf{Expend})$	itures $> 0$ )
	Days		<b>P</b> (Employed at Shock Firm)		Days		Mammography	
Gender of Coworker & Shock Worker	Same (1)	Different (2)	$\frac{\text{Same}}{(3)}$	Different (4)	Same (5)	Different (6)	Same (7) 0.0091** (0.0039)	Different (8) 0.0010 (0.0067)
Health Shock $\times$ Post	$\begin{array}{c} 0.8302 \\ (0.7355) \end{array}$	1.7573 (1.0697)	$0.0282^{*}$ (0.0158)	0.0111 (0.0225)	-0.8885 (0.5477)	-1.2507 (0.7975)		
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	67,109 1,338 0.258 88.459 0.938	$25,949 \\ 749 \\ 0.217 \\ 89.304 \\ 1.968$	67,109 1,338 0.698 0.920 3.068	$25,949 \\749 \\0.348 \\0.928 \\1.200$	67,109 1,338 0.189 7.851 -11.316	25,949 749 0.674 7.775 -16.086	34,285 692 0.368 0.068 13.393	$13,765 \\ 436 \\ 0.625 \\ 0.060 \\ 1.629$

Table A.10 Average Effects of Health Shocks on Coworkers' Outcomes by Gender of Coworker & Shock Worker

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' outcomes for combinations of the gender of the coworkers and the shock worker. We estimate Equation (1) separately for coworkers with the same gender as the shock worker and coworkers with a different gender as the shock worker. Columns (1), (3), (5), and (7) report the results for the first group, Columns (2), (4), (6), and (8) those for the second group. Columns (1) and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. In Columns (7) and (8), we define a binary indicator that equals one if the outpatient expenditures for mammography in a given quarter are larger than zero and zero otherwise. This outcome is analyzed only for women. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Emplo	oyment		Retire	ment	$\mathbf{P}(\mathbf{Expend})$	itures $> 0$ )
	Days		P(Em at Shoc	ployed k Firm)	Da	ys	Mammography	
Type of Shock	Cancer (1)	Cardio (2)	Cancer (3)	Cardio (4)	Cancer (5)	Cardio (6)	Cancer (7)	Cardio (8)
Health Shock $\times$ Post	1.2147 (0.7458)	0.7000 (1.0749)	$0.0164 \\ (0.0167)$	$0.0305 \\ (0.0235)$	$-1.3900^{**}$ (0.5736)	$0.0182 \\ (0.7021)$	$0.0088^{**}$ (0.0040)	-0.0022 (0.0066)
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	63,236 1,105 0.050 88.591 1.371	$29,821 \\ 504 \\ 0.523 \\ 88.948 \\ 0.787$	63,236 1,105 0.617 0.916 1.792	$29,821 \\ 504 \\ 0.800 \\ 0.934 \\ 3.264$	63,236 1,105 0.584 8.504 -16.346	$29,821 \\ 504 \\ 0.795 \\ 6.361 \\ 0.286$	35,679 809 0.468 0.065 13.486	$12,370 \\ 322 \\ 0.892 \\ 0.066 \\ -3.382$

 Table A.11
 Average Effects of Health Shocks on Coworkers' Outcomes by Type of Shock

*Note* — The table reports the estimated coefficients for the effect of a health shock on coworkers' outcomes by the type of severe health shock. We estimate Equation (1) separately for health shocks due to a cancer diagnosis and health shocks due to a cardiovascular disease. Columns (1), (3), (5), and (7) report the results for the first group, Columns (2), (4), (6), and (8) those for the second group. Columns (1) and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. In Columns (7) and (8), we define a binary indicator that equals one if the outpatient expenditures for mammography in a given quarter are larger than zero and zero otherwise. This outcome is analyzed only for women. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		Emplo	yment		Retire	Retirement		Wages			
	Days		<b>P</b> (Employed at Shock Firm)		Da	Days		Log Daily Wage		Wage Growth	
Shock Worker Leaves Firm?	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	
Health Shock $\times$ Post	$2.0715^{**}$ (0.9329)	$\begin{array}{c} 0.0172 \\ (0.7916) \end{array}$	$0.0441^{*}$ (0.0231)	-0.0014 (0.0153)	$-1.4503^{**}$ (0.6566)	-0.5479 (0.6267)	-0.0040 (0.0105)	0.0066 (0.0055)	-0.0015 (0.0029)	0.0005 (0.0010)	
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
N	45,785	47,273	45,785	47,273	45,785	47,273	41,362	42,766	39,037	40,444	
# Clusters	827	782	827	782	827	782	826	782	826	782	
p-Value of Pre-Coefficients	0.440	0.497	0.816	0.433	0.073	0.797	0.932	0.199	0.779	0.685	
Uutcome Mean	88.025	89.379	0.911	0.933	8.275	7.371	4.115	4.145	0.005	0.007	
Effect in $\gamma_0$ of Mean	2.353	0.019	4.837	-0.148	-11.528	-1.433	-0.096	0.160	-21.070	1.478	

Table A.12 Average Effects of Health Shocks on Coworkers' Labor Market Outcomes by Whether Shock Worker Leaves the Firm

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' labor market outcomes by whether the shock worker leaves the shock firm after the health shock. We estimate Equation (1) separately for coworkers of shock workers who leave the shock firm within two quarters after the health shock and for coworkers of shock workers who do not leave the shock firm in this time period. Columns (1), (3), (5), (7), and (9) report the results for the first group, Columns (2), (4), (6), (8), and (10) those for the second group. Columns (1) and (2) show the results for the number of employment days in a given quarter. Columns (3) and (4) use a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Columns (5) and (6) report the results for the quarterly number of retirement days. Columns (7) and (8) use the log daily wage as an outcome. Columns (9) and (10) consider wage growth, which is defined as the difference in log daily wages between two consecutive quarters, as an outcome. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\*\* p < 0.05, \*\*\*\* p < 0.001.

	Sick Leave Days				$\mathbf{P}(\text{Expenditures} > 0)$						
	All Causes		Easy-t Diag	Easy-to-Fake Diagnoses		General Health Check-Up		PSA Screening		Mammography	
Shock Worker Leaves Firm?	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	
Health Shock $\times$ Post	-0.1164 (0.2566)	-0.0732 (0.2777)	-0.0278 (0.0409)	$\begin{array}{c} 0.0375 \ (0.0380) \end{array}$	$\begin{array}{c} 0.0014 \\ (0.0030) \end{array}$	$\begin{array}{c} 0.0007 \\ (0.0031) \end{array}$	$0.0066 \\ (0.0061)$	$\begin{array}{c} 0.0032\\ (0.0062) \end{array}$	$0.0058 \\ (0.0048)$	0.0072 (0.0047)	
Individual FE Year $\times$ Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
N # Clusters	$45,013 \\ 827$	$46,\!849$ 782	$44,992 \\ 827$	$46,838 \\ 782$	$44,989 \\ 827$	$46,840 \\ 782$	$22,395 \\ 492$	$21,393 \\ 467$	$22,598 \\ 565$	$25,\!451 \\ 567$	
p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	0.052 2.424 -4.802	0.270 2.459 -2.979	$0.715 \\ 0.134 \\ -20.778$	$\begin{array}{c} 0.728 \\ 0.103 \\ 36.365 \end{array}$	$\begin{array}{c} 0.399 \\ 0.039 \\ 3.736 \end{array}$	$0.328 \\ 0.042 \\ 1.604$	$\begin{array}{c} 0.126 \\ 0.058 \\ 11.467 \end{array}$	$0.753 \\ 0.062 \\ 5.201$	$0.597 \\ 0.063 \\ 9.062$	$0.485 \\ 0.067 \\ 10.745$	

Table A.13 Average Effects of Health Shocks on Coworkers' Health Outcomes by Whether Shock Worker Leaves the Firm

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' health outcomes by whether the shock worker leaves the shock firm after the health shock. We estimate Equation (1) separately for coworkers of shock workers who leave the shock firm within two quarters after the health shock and for coworkers of shock workers who do not leave the shock firm in this time period. Columns (1), (3), (5), (7), and (9) report the results for the first group, Columns (2), (4), (6), (8), and (10) those for the second group. Columns (1) and (2) show the results for the number of sick leave days in a given quarter. Columns (3) and (4) include only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). In Columns (5) and (6), we define a binary indicator that equals one if the outpatient expenditures for the general health screening program in a given quarter are larger than zero and zero otherwise. Columns (7) and (8) as well as (9) and (10) use analogous binary indicators for expenditures related to PSA tests and mammography. These outcomes are analyzed only for men and women, respectively. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.

		L	abor Market O	utcomes		Health Outcomes					
	En	nployment	Retirement	Wag	Wages		Sick Leave Days		$\mathbf{P}(\text{Expenditures} > 0)$		
	Days (1)	P(Employed at Shock Firm) (2)	Days (3)	Log Daily Wage (4)	Wage Growth (5)	All Causes (6)	Easy-to-Fake Diagnoses (7)	General Health Check-Up (8)	PSA Screening (9)	Mammography (10)	
Health Shock $\times$ Post	$1.2434^{*}$ (0.7472)	$0.0263^{*}$ (0.0135)	$-1.1454^{**}$ (0.5561)	0.0084 (0.0056)	$\begin{array}{c} 0.0015\\ (0.0012) \end{array}$	-0.0768 (0.2181)	$0.0265 \\ (0.0311)$	$0.0005 \\ (0.0026)$	0.0061 (0.0052)	0.0055 (0.0039)	
Individual FE Year × Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	$93,058 \\ 1,605 \\ 0.388 \\ 88.613 \\ 1.403$	$93,058 \\ 1,605 \\ 0.583 \\ 0.920 \\ 2.859$	$93,058 \\ 1,605 \\ 0.522 \\ 8.263 \\ -13.862$	$\begin{array}{c} 84,129\\ 1,604\\ 0.885\\ 4.100\\ 0.206\end{array}$	$79,482 \\ 1,604 \\ 0.881 \\ 0.006 \\ 23.641$	$91,862 \\ 1,605 \\ 0.244 \\ 2.253 \\ -3.407$	91,830 1,605 0.508 0.111 23.895	$91,829 \\ 1,605 \\ 0.207 \\ 0.041 \\ 1.241$	$\begin{array}{r} 43,788\\958\\0.219\\0.060\\10.216\end{array}$	$\begin{array}{c} 48,050\\ 1,128\\ 0.522\\ 0.064\\ 8.502 \end{array}$	

Table A.14 Average Effects of Health Shocks on Coworkers' Outcomes – Weighted Regressions

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' labor market and health outcomes. The results are based on Equation (1) where the regression is weighted by the inverse number of workers employed at the shock firm in the quarter of the (placebo) health shock. Column (1) shows the results for the number of employment days in a given quarter. Column (2) uses a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Column (3) reports the results for the quarterly number of retirement days. Column (4) uses the log daily wage as an outcome. Column (5) considers wage growth, which is defined as the difference in log daily wages between two consecutive quarters, as an outcome. Column (6) shows the results for the number of sick leave days in a given quarter. Column (7) includes only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). In Column (8), we define a binary indicator that equals one if the outpatient expenditures for the general health screening program in a given quarter are larger than zero and zero otherwise. Columns (9) and (10) use analogous binary indicators for expenditures related to PSA tests and mammography. These outcomes are analyzed only for men and women, respectively. The sample period comprises the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome wariable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment peri

		L	abor Market O	utcomes		Health Outcomes					
	En	nployment	Retirement	Wages		Sick Leave Days		$\mathbf{P}(\text{Expenditures} > 0)$			
	Days (1)	P(Employed at Shock Firm) (2)	Days (3)	Log Daily Wage (4)	Wage Growth (5)	All Causes (6)	Easy-to-Fake Diagnoses (7)	General Health Check-Up (8)	PSA Screening (9)	Mammography (10)	
Health Shock $\times$ Post	$1.0480^{*}$ (0.6356)	0.0163 (0.0145)	$(0.4394)^{-1.0249**}$	0.0017 (0.0057)	-0.0003 (0.0014)	-0.1227 (0.1884)	$\begin{array}{c} 0.0115\\ (0.0281) \end{array}$	0.0001 (0.0021)	0.0039 (0.0043)	$0.0063^{*}$ (0.0033)	
Individual FE Year × Birth Year FE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	
N # Clusters p-Value of Pre-Coefficients Outcome Mean Effect in % of Mean	99,412 1,662 0.024 87.283 1.201	$99,412 \\ 1,662 \\ 0.057 \\ 0.882 \\ 1.851$	$99,412 \\ 1,662 \\ 0.280 \\ 7.685 \\ -13.336$	$89,155 \\ 1,661 \\ 0.328 \\ 4.126 \\ 0.041$	84,047 1,661 0.469 0.006 -4.224	97,541 1,662 0.026 2.546 -4.820	$97,481 \\ 1,662 \\ 0.256 \\ 0.128 \\ 8.955$	$97,479 \\ 1,662 \\ 0.275 \\ 0.040 \\ 0.241$	$\begin{array}{c} 46,829\\997\\0.144\\0.059\\6.608\end{array}$	50,663 1,177 0.600 0.066 9.529	

Table A.15 Average Effects of Health Shocks on Coworkers' Outcomes – Relaxed Employment Constraint

Note — The table reports the estimated coefficients for the effect of a health shock on coworkers' labor market and health outcomes. The results are based on Equation (1). In contrast to the main estimation sample, where coworkers and shock workers have to be employed for at least four quarters before the health shock (including the quarter of the health shock), we only require coworkers and shock workers to be employed in the shock firm in the quarter of the shock for this estimation sample. Column (1) shows the results for the number of employment days in a given quarter. Column (2) uses a binary indicator that equals one if a coworker is employed at the firm where the (placebo) health shock occurred and zero otherwise. Column (3) reports the results for the quarterly number of retirement days. Column (4) uses the log daily wage as an outcome. Column (5) considers wage growth, which is defined as the difference in log daily wages between two consecutive quarters, as an outcome. Column (6) shows the results for the number of sick leave days in a given quarter. Column (7) includes only sick leaves associated with easy-to-fake diagnoses, including the common cold (ICD-10 J04 and J06), lower back pain (ICD-10 M54.5), and headache (ICD-10 R51). In Column (8), we define a binary indicator that equals one if the outpatient expenditures for the general health screening program in a given quarter are larger than zero and zero otherwise. Columns (9) and (10) use analogous binary indicators for expenditures related to PSA tests and mammography. These outcomes are analyzed only for men and women, respectively. The sample period comprises the 12 quarters preceding and the 12 quarters (including the quarter of the health shock) following the (placebo) health shock. The number of observations, the number of clusters, the p-value of an F-test for the joint significance of all pre-treatment coefficients of Equation (2), the mean of the outcome variable for the control group, and the coefficient in percent of the outcome mean are reported at the bottom of the table. For all outcomes except retirement, the outcome mean is based on observations from the pre-treatment period. The outcome mean for retirement is based on the post-treatment period. Standard errors are clustered at the shock firm level. \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.001.